% version of 10 April 2014
\documentclass[11pt]{amsart}
\usepackage{graphicx, color}
\usepackage{amscd}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{mathrsfs}
\textwidth=6in \textheight=9.5in \topmargin=-0.5cm
\oddsidemargin=0.5cm \evensidemargin=0.5cm
%\usepackage[notref,notcite]{showkeys}

\newtheorem{theorem}{Theorem}
\newtheorem{ex}{Example}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{claim}{Claim}
\newtheorem{step}{Step}


\numberwithin{equation}{section}
\newcommand{\RR}{\mathbb R}
\newcommand{\NN}{\mathbb N}
\newcommand{\PP}{\mathbb P}
\renewcommand{\le}{\leqslant}
\renewcommand{\leq}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\geq}{\geqslant}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\baselineskip=16pt plus 1pt minus 1pt

\begin{document}
\begin{thebibliography}{99}
\bibitem{abdellaoui} {\sc B. Abdellaoui, V. Felli and I. Peral},
{\em Existence and nonexistence results for quasilinear elliptic equations involving the
$p$-Laplacian}, Boll. Unione Mat. Ital. Sez. B Artic. Ric. Mat. (8) \textbf{9} (2006), no. 2, 445-484.

\bibitem{AT} {\sc S. Alama and G. Tarantello}, {\em Elliptic problems with nonlinearities indefinite in sign}, J. Funct. Anal. \textbf{141} (1) (1996), 159-215.

\bibitem{AT2} {\sc S. Alama and G. Tarantello}, {\em On semilinear elliptic equations with indefinite nonlinearities}, Calc. Var. Partial Differential Equations \textbf{1} (4) (1993), 439-475.

 \bibitem{ABC} {\sc A. Ambrosetti, H. Brezis, and G. Cerami}, {\em Combined effects of concave and convex nonlinearities in some elliptic problems}, J. Funct. Anal. \textbf{122} (2) (1994), 519-543.

\bibitem{ambroazo} {\sc A. Ambrosetti, J. Garcia Azorero and I. Peral}, {\em
Multiplicity results for some nonlinear elliptic equations}, J. Funct. Anal. \textbf{137} (1996), no.~1, 219-242.

\bibitem{bartschwillem} {\sc  T. Bartsch and M. Willem}, {\em On an elliptic equation with concave and convex nonlinearities}, Proc. Amer. Math. Soc. \textbf{123} (1995), no.~11, 3555-3561.

\bibitem{BCDN} {\sc H. Berestycki, I. Capuzzo Dolcetta and L. Nirenberg}, {\em Variational methods for indefinite superlinear homogeneous elliptic problems}, NoDEA Nonlinear Differential Equations Appl. \textbf{2} (4) (1995), 553-572.

\bibitem{boccardo} {\sc L. Boccardo}, {\em A Dirichlet problem with singular and supercritical nonlinearities}, Nonlinear Anal. \textbf{75} (2012), no.~12, 4436-4440.

\bibitem{brezis} {\sc H. Brezis}, {\em Functional Analysis, Sobolev Spaces and Partial Differential Equations}, Universitext, Springer, New York, 2011.

\bibitem{charro} {\sc F. Charro, E. Colorado and I. Peral}, {\em Multiplicity of solutions to uniformly elliptic fully nonlinear equations with concave-convex
right-hand side}, J. Differential Equations \textbf{246} (2009), no.~11, 4221-4248.

\bibitem{ciarlet}  {\sc P.G. Ciarlet}, {\em Linear and Nonlinear Functional Analysis with Applications},
Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2013.

\bibitem{CGR} {\sc F. C\^{\i}rstea, M. Ghergu and V. R\u{a}dulescu}, {\em Combined effects of asymptotically linear and singular nonlinearities in bifurcation problems of
Lane-Emden-Fowler type}, J. Math. Pures Appl. \textbf{84} (2005), 493-508.

\bibitem{Cola} {\sc F. Colasuonno}, {\em Some problems involving the $p(x)$-polyharmonic Kirchhoff operator}, Ph.D. Thesis, University of Bari (2012).

\bibitem{denapoli} {\sc P. De N\'apoli and M.C. Mariani}, {\em Mountain pass solutions to equations of $p$-Laplacian type}, Nonlinear Anal. \textbf{54} (2003), no. 7, 1205-1219.

\bibitem{CPV} {\sc F. Colasuonno, P. Pucci and Cs. Varga}, {\em Multiple solutions for an eigenvalue
problem involving $p$-Laplacian type operators}, Nonlinear Anal. \textbf{75} (2012), 4496-4512.

\bibitem{DFGU} {\sc D.G. De Figueiredo, J.-P. Gossez and P. Ubilla}, {\em Multiplicity results for a family of semilinear elliptic problems under local superlinearity and
sublinearity}, J. Eur. Math. Soc. (JEMS) \textbf{8} (2) (2006), 269-286.

\bibitem{DFGU2} {\sc D.G. De Figueiredo, J.-P. Gossez and P. Ubilla}, {\em Local superlinearity and sublinearity for indefinite semilinear elliptic problems}, J. Funct. Anal. \textbf{199} (2) (2003), 452-467.

\bibitem{azoreroperal} {\sc J. Garcia Azorero and I. Peral Alonso}, {\em Multiplicity of solutions for elliptic problems with critical exponent or with a nonsymmetric
term}, Trans. Amer. Math. Soc. \textbf{323} (1991), no.~2, 877-895.

\bibitem{azoreroperal2} {\sc J. Garcia Azorero and I. Peral Alonso}, {\em
Some results about the existence of a second positive solution in a quasilinear critical
problem}, Indiana Univ. Math. J. \textbf{43} (1994), no.~3, 941-957.

\bibitem{ghoussoub} {\sc N. Ghoussoub and C. Yuan}, {\em Multiple solutions for quasi-linear PDEs involving the critical Sobolev and Hardy exponents}, Trans. Amer. Math. Soc. \textbf{352} (2000), no.~12, 5703-5743.

\bibitem{KM} {\sc A. Krist\'aly and Gh. Moro\c{s}anu}, {\em New competition phenomena in Dirichlet problems}, J. Math. Pures Appl. (9) \textbf{94} (2010), 6, 555-570.

%\bibitem{KLV} {\sc A. Krist\'aly, H. Lisei and Cs. Varga}, {\em Multiple solutions for $p$-Laplacian type %equations}, Nonlinear Anal. \textbf{68} (2008), 1375-1381.

%\bibitem{KMT} {\sc A. Krist\'aly, Gh. Moro\c{s}anu and S. Tersian}, {\em Quasilinear elliptic problems in %$\RR^N$ involving oscillatory nonlinearities}, J. Differential Equations \textbf{235} (2) (2007), 366-375.

\bibitem{MM} {\sc M. Marcus and V. Mizel}, {\em Every superposition operator mapping one Sobolev space into another is continuous}, J. Funct. Anal. \textbf{33} (1979), 217-229.

\bibitem{MR} {\sc G. Molica Bisci and D. Repov\v{s}}, {\em Multiple solutions for elliptic equations involving a general operator in divergence form}, submitted for publication.

\bibitem{doo} {\sc J.M. do \'O and U. Severo}, {\em Quasilinear Schr\"odinger equations involving concave and convex nonlinearities}, Commun. Pure Appl. Anal. \textbf{8} (2009), no.~2, 621-644.

\bibitem{OO} {\sc F. Obersnel and P. Omari}, {\em Positive solutions of elliptic problems with locally oscillating nonlinearities}, J. Math. Anal. Appl. \textbf{323} (2) (2006), 913-929.

\bibitem{OZ} {\sc P. Omari and F. Zanolin}, {\em Infinitely many solutions of a quasilinear elliptic problem with an oscillatory potential}, Comm. Partial Differential Equations \textbf{21} (1996), 721-733.

\bibitem{puccirad}  {\sc P. Pucci and V. R\u{a}dulescu}, {\em Combined effects in quasilinear elliptic problems with lack of compactness}, Atti Accad. Naz. Lincei Cl. Sci. Fis. Mat. Natur. Rend. Lincei (9) Mat. Appl. \textbf{22} (2011), no.~2, 189-205.

\bibitem{PS1} {\sc P. Pucci and R. Servadei}, {\em On weak solutions for p-Laplacian equations with weights}, Atti Accad. Naz. Lincei Cl. Sci. Fis. Mat. Natur. Rend. Lincei (9) Mat. Appl. \textbf{18} (2007), 257-267.

\bibitem{PS2} {\sc P. Pucci and R. Servadei}, {\em Existence, non-existence and regularity of radial ground states for p- Laplacian equations with singular weights}, Ann. Inst. H. Poincar�e Anal. Non Lin\'eaire \textbf{25} (2008), 505-537.

\bibitem{SR} {\sc J. Saint Raymond}, {\em On the multiplicity of the solutions of the equation $-\Delta u = \lambda f(u)$}, J. Differential Equations \textbf{180} (2002), 65-88.

\end{thebibliography}

%\hfill\today\bigskip

\title[Competition phenomena for elliptic equations]{Competition phenomena for elliptic equations involving a general operator in divergence form}

\thanks{The first author was supported by the GNAMPA Project 2012 {\it Esistenza e molteplicit\`{a} di soluzioni per problemi differenziali non lineari}, the second one by the Grant CNCS PCE--47/2011, while   the third author was supported by the MIUR National Research Project
{\it Variational and Topological Methods in the Study of Nonlinear
Phenomena} and by the ERC grant $\epsilon$ ({\it Elliptic Pde's and
Symmetry of Interfaces and Layers for Odd Nonlinearities}).}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\author[G. Molica Bisci]{Giovanni Molica Bisci}
\address{Dipartimento PAU,
          Universit\`a `Mediterranea' di Reggio Calabria,
          Via Melissari 24, 89124 Reggio Calabria, Italy}
\email{\tt gmolica@unirc.it}

\author[V. R\u{a}dulescu]{Vicen\c{t}iu D. R\u{a}dulescu}
\address{Department of Mathematics, Faculty of Science, King Abdulaziz University,
Jeddah, Saudi Arabia \&
           Institute of Mathematics `Simion Stoilow' of the Romanian Academy,
          014700 Bucharest, Romania}
\email{\tt vicentiu.radulescu@math.cnrs.fr}

\author[R. Servadei]{Raffaella Servadei}
\address{Dipartimento di Matematica e Informatica,
          Universit\`a della Calabria,
          Ponte Pietro Bucci 31 B, 87036 Arcavacata di Rende (Cosenza), Italy}
\email{\tt servadei@mat.unical.it}

\keywords{Dirichlet problems, elliptic operators, $p$-Laplacian operator, infinitely many solutions, variational methods.\\
\phantom{aa} 2010 AMS Subject Classification: Primary:
35J62, 35J92, 35J20;
Secondary: 35J62, 35J15, 47J30.}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\begin{abstract}
In this paper, by using variational methods, we study the following elliptic problem
$$\left\{\begin{array}{ll}
-{\rm div}A(x,\nabla u)= \lambda \beta(x) u^q + f(u) &  \mbox{in}\,\,\, \Omega\\
u\geq 0 &  \mbox{in}\,\,\, \Omega\\
u=0 &  {\rm on}\,\,\, \partial \Omega
 \end{array}\right.
$$
involving a general operator in divergence form of $p$-Laplacian type ($p>1$). In our context $\Omega$ is a bounded domain of $\RR^N$, $N\geq 3$, with smooth boundary $\partial \Omega$, $A$ is a continuous function with potential~$a$, $\lambda$ is a real parameter, $\beta\in L^\infty(\Omega)$ is allowed to be indefinite in sign, $q>0$ and  $f:[0,+\infty)\to \RR$ is a continuous function oscillating near the origin or at infinity.

Through variational and topological methods we show that the number of solutions of the problem is influenced by the competition between the power $u^q$ and the oscillatory term $f$. To be precise, we prove that, when $f$ oscillates near the origin, the problem admits infinitely many solutions when $q\geq p-1$ and at least a finite number of solutions when $0<q<p-1$. While, when $f$ oscillates at infinity, the converse holds true, that is, there are infinitely many solutions if $0<q\leq p-1$, and at least a finite number of solutions if $q>p-1$. In all these cases we also give some estimates for the $W^{1,p}$ and $L^\infty$--norm of the solutions.

The results presented here extend some recent contributions obtained for equations driven by the Laplace operator, to the case of the $p$-Laplacian  or even to more general differential operators.
\end{abstract}

\maketitle

\tableofcontents


\section{Introduction}\label{sec:introduzione}
Competition phenomena in elliptic equations have been widely studied in the literature in different contexts. After the seminal work~\cite{ABC}, where Ambrosetti, Brezis and Cerami studied a Laplacian equation involving a concave-convex nonlinearity, a lot of papers appeared on this subject (see, for instance, \cite{ambroazo, bartschwillem, charro, DFGU, DFGU2, azoreroperal, azoreroperal2, doo, puccirad} and the references therein).
Also when dealing with singular terms, the interactions with different type of nonlinearities were investigated: see, for instance, \cite{boccardo} for supercritical nonlinearities, \cite{ghoussoub, PS1, PS2} for equations involving superlinear and subcritical terms, \cite{abdellaoui} for the concave-convex setting and \cite{CGR} for the asymptotically linear case, just to name a few.

Equations driven by the $p$-Laplace operator, or, more generally, by operators in divergence form of $p$-Laplacian type were widely studied recently in the literature (see, e.g. \cite{abdellaoui, charro, denapoli, CPV, MR} and the references therein).

In this paper we are interested in problems driven by general operators of $p$-Laplacian type involving oscillatory terms, in presence of a concave or convex power. Usually,  equations involving oscillatory nonlinearities give infinitely many distinct solutions (see, e.g., the papers \cite{OO,OZ,SR} and references therein for more details), but the presence of an additional term may alter the situation. For instance, in \cite{KM} the authors studied a Laplacian equation with an oscillatory term in presence of a power and they showed that the number of solutions depend strongly on this power: when there is an oscillatory term near the origin the equation under consideration admits infinitely many distinct solutions if the power is convex, while it has a finite number of distinct solution when the power is concave. In the case of oscillations at infinity, the converse result holds true.

The aim of the present paper is to extend some of the results obtained in \cite{KM} to a general class of quasilinear equations of $p$--Laplacian type. Precisely, here we deal with the following problem
\begin{equation}\label{problem}
\left\{ \begin{array}{ll}
-{\rm div}A(x,\nabla u) = \lambda \beta(x) u^q + f(u) & \mbox{in} \, \Omega\\ u\geq 0 & \mbox{in} \, \Omega\\ u=0 & \mbox{on} \, \partial \Omega, \end{array} \right.
\end{equation}
where $\Omega\subset \RR^N$, $N \geq 3$, is a bounded domain with smooth boundary $\partial \Omega$, $q>0$ and $\lambda\in \RR$ are parameters, while $\beta \in L^{\infty}(\Omega)$ and $f:[0,+\infty)\to \RR$ is a continuous function.

We also assume that $A:\overline \Omega \times \RR^N\to \RR^N$ is a function such that
\begin{equation}\label{Acontinua}
A\,\,\, \mbox{is continuous in}\,\,\,\overline\Omega\times \RR^N\,;
\end{equation}
\begin{equation}\label{crescita}
\begin{aligned}
& \mbox{there exist}\,\,\, p>1\,\,\, \mbox{and two positive constants}\,\,\, \Gamma_1\leq \Gamma_2\,\,\, \mbox{such that}\\
& A(x,\xi)\cdot \xi\geq \Gamma_1 |\xi|^p\,\,\,\,\,\,\,\,\mbox{and}\,\,\,\,\,\,\,\,|A(x,\xi)|\leq \Gamma_2|\xi|^{p-1}\,\,\, \mbox{for all}\, (x,\xi)\in \Omega\times\RR^N\,;
\end{aligned}
\end{equation}
\begin{equation}\label{potenziale}
A=\nabla_\xi a\,.
\end{equation}
Here, $a:\overline\Omega\times \RR^N\to \RR$ is the potential of $A$ (with respect to the second variable) and satisfies the following conditions:
\begin{equation}\label{acontinua}
a\,\,\, \mbox{is continuous in}\,\,\,\overline\Omega\times \RR^N\,;
\end{equation}
\begin{equation}\label{condA}
a(x,0)=0\,\,\, \mbox{and}\,\,\, a(x,\xi)=a(x,-\xi)\,\,\, \mbox{for all}\, (x, \xi)\in\Omega\times \RR^N\,;
\end{equation}
\begin{equation}\label{Aconvex}
a(x,\cdot)\,\,\, \mbox{is strictly convex in}\,\,\, \RR^N\,\,\, \mbox{for all}\, x\in \Omega\,.
\end{equation}

Assumptions~\eqref{Acontinua}--\eqref{Aconvex} are natural structural conditions. As a model for $A$ we can take the function
$$A(x, \xi)=|\xi|^{p-2}\xi\,,$$
(of course, in this case $a(x,\xi)=|\xi|^p/p$) which gives rise to the well-known $p$--Laplace operator~$\Delta_p$, defined as
$$\Delta_p u:={\rm div}(|\nabla u|^{p-2}\nabla u)\,.$$

The purpose of this paper is to study the number and the behavior of the solutions of problem~\eqref{problem}, when $f$ \emph{oscillates near the origin} or \emph{at infinity}. This analysis will be carried on using variational and topological techniques. In the sequel, we state our main results, treating separately the two cases, that is, when the nonlinearity $f$ oscillates near the origin or at infinity, respectively.

Finally, we would like to emphasize that the coefficient $\beta\in L^{\infty}(\Omega)$ in problem~\eqref{problem} is allowed to be \emph{indefinite} in sign, as suggested by several well-known works (see, for instance, \cite{AT, AT2, BCDN, DFGU, DFGU2} and references therein).

The plan of the paper is as follows. In Section~\ref{sec:main} we will state the main results of the paper in the two different situations when $f$ oscillates near the origin or at infinity. In Section~\ref{sec:comments} we will comment the assumptions on the data of problem~\eqref{problem}. In Section~\ref{sec:auxiliary} we will consider an auxiliary problem and for it we will prove the existence of solutions by direct minimization. Finally, in Section~\ref{sec:origin} we will study problem~\eqref{problem} in presence of an oscillation term near zero, while Section~\ref{sec:infty} is devoted to the case of oscillations at infinity.

We refer to the recent books by Haim Brezis \cite{brezis} and
Philippe G.~Ciarlet \cite{ciarlet} for related results and complements.



\section{Main results}\label{sec:main}
This section is devoted to the main results of the paper, where we prove the existence of infinitely many solutions for problem~\eqref{problem} in these two different contexts:

\smallskip
\noindent
(i) $f$ oscillating near the origin and $q\geq p-1$;\\
(ii) $f$ oscillating at infinity and $0<q\leq p-1$\,,

\smallskip
\noindent
while, in the remaining cases, that is, when

\smallskip
\noindent
(iii) $f$ oscillates near the origin and $0<q< p-1$;\\
(iv) $f$ oscillates at infinity and $q> p-1$\,,

\smallskip
\noindent
we show the existence of at least a finite number of solutions. Here $p$ is the parameter appearing in \eqref{crescita}.

In all these cases we assume that $f:[0, +\infty)\to \RR$ is a continuous function. Also, we denote by $F$ the function
\begin{equation}\label{F}
F(s) :=\int^s_0 f(t)\,dt
\end{equation}
for any $s>0$\,.

As usual, here and in the sequel, $W_{0}^{1,p}(\Omega)$ will denote the closure of $C^{\infty}_0(\Omega)$ with respect to the norm
$$
\|u\|_{W^{1,p}_0(\Omega)}:= \Big(\int_\Omega |\nabla u(x)|^p dx\Big)^{1/p}.
$$


\subsection{Oscillation near the origin}\label{subsec:origin}
In this framework we assume that the following conditions are satisfied:
\begin{equation}\label{f2origin}
\displaystyle \liminf_{s\to 0^+} \frac{f(s)}{s^{p-1}}=:-\ell_0\in [-\infty, 0)\,;
\end{equation}
\begin{equation}\label{f1origin}
\displaystyle -\infty < \liminf_{s\to 0^+} \frac{F(s)}{s^p}\leq \limsup_{s\to 0^+} \frac{F(s)}{s^p} = +\infty\,,
\end{equation}
where $p$ is the parameter given in \eqref{crescita} and $F$ is the function defined in \eqref{F}\,.

As a model for $f$ we can take the function
$$f(s)=\left\{\begin{array}{ll}
\alpha s^{\alpha-1}\Big(1-\sin s^{-\sigma}\Big)+\sigma s^{\alpha-\sigma-1} \cos s^{-\sigma}-p\gamma s^{p-1} & \mbox{if}\,\,\, s>0\\
0 & \mbox{if}\,\,\, s=0\,,
\end{array}\right.$$
where $\alpha$, $\sigma$ and $\gamma$ are such that $1<\sigma+1<\alpha<p$ and $\gamma>0$.
Note that $f$ is continuous in $[0, +\infty)$ and $F$ is the following function
$$F(s)=\int_0^s f(t)\,dt=s^\alpha \Big(1-\sin s^{-\sigma}\Big)-\gamma s^p,\,\,\, s>0\,.$$

Another prototype for $f$ is given by
$$f(s)=\left\{\begin{array}{ll}
\alpha s^{\alpha-1}\cos^2 s^{-\sigma}-2\sigma s^{\alpha-\sigma-1} \cos s^{-\sigma}\sin s^{-\sigma}-p\gamma s^{p-1} & \mbox{if}\,\,\, s>0\\
0 & \mbox{if}\,\,\, s=0\,,
\end{array}\right.$$
where $\alpha$, $\sigma$ and $\gamma$ are such that $1<\alpha<p$, $\sigma>0$, $\alpha-\sigma>1$  and $\gamma>0$.
Thanks to these choices of the parameters $f$ is continuous in $[0, +\infty)$. Also $F$ is the following function
$$F(s)=\int_0^s f(t)\,dt=s^\alpha \cos^2 s^{-\sigma}-\gamma s^p,\,\,\, s>0\,.$$


In both examples, we deduce by direct calculations that $f$ and $F$ satisfy assumptions \eqref{f2origin} and \eqref{f1origin}.

\medskip

In this setting our main result can be stated as follows.
\begin{theorem}\label{thorigin}
Let $\Omega\subset \RR^N$, $N \geq 3$, be a bounded domain with smooth boundary, $\lambda\in \RR$ and let $A:\overline \Omega \times \RR^N\to \RR^N$ and $a:\overline\Omega\times \RR^N\to \RR$ be two functions satisfying \eqref{Acontinua}--\eqref{potenziale} and \eqref{acontinua}--\eqref{Aconvex}, respectively.
Assume that $\beta\in L^{\infty}(\Omega)$ and $f \in C([0,+\infty);\RR)$ satisfies \eqref{f2origin} and \eqref{f1origin}. If either
\begin{enumerate}
\item[$ a)$] $q = p-1$, $\ell_0\in (0, +\infty)$ and $\lambda \beta(x) < \lambda_0$ a.e. $x\in \Omega$ for some $\lambda_0\in (0, \ell_0)$ or
\item[$ b)$] $q = p-1$, $\ell_0=+\infty$ and $\lambda \in \RR$ is arbitrary or
\item[$ c)$] $q >p-1$ and $\lambda \in \RR$ is arbitrary,
\end{enumerate}
then there exists a sequence $\{u_j\}_j$ in $W_0^{1,p}(\Omega)$ of distinct weak solutions of problem~\eqref{problem} such that
\begin{equation} \label{1.1.0}
\lim_{j\to +\infty} \|u_j\|_{W^{1,p}_0(\Omega)}
= \lim_{j\to +\infty} \|u_j\|_{L^{\infty}(\Omega)}=0.
\end{equation}

While, if $0<q<p-1$, then for every $k\in \NN$ there exists $\Lambda_k>0$ such that problem~\eqref{problem} has at least $k$ distinct weak solutions $u_1, \dots, u_k\in W^{1,p}_0(\Omega)$
such that
\begin{equation} \label{1.1.0bis}
\|u_j\|_{W^{1,p}_0(\Omega)}\leq 1/j \qquad \mbox{and} \qquad \|u_j\|_{L^{\infty}(\Omega)}\leq 1/j\,, \,\,\,\,\, j=1,\dots, k
\end{equation}
provided $|\lambda|<\Lambda_k$.
\end{theorem}

Assumption~\eqref{f2origin} yields the existence of solutions for problem~\eqref{problem}, while \eqref{f1origin} allows us to deduce some information about the number of the solutions.

We also would like to note that assertion~$b)$ covers also the case when the power $q$ is critical or supercritical, that is the case when $q\geq p^*$, where
\begin{equation}\label{pstar}
p^*=Np/(N-p),\,\,\, N>p
\end{equation}
is the Sobolev critical exponent.

\subsection{Oscillation at infinity}\label{subsec:infty}
In this framework we assume that the following assumptions hold true:
\begin{equation}\label{f2infty}
\displaystyle\liminf_{s\to +\infty} \frac{f(s)}{s^{p-1}}=:-\ell_\infty\in[-\infty, 0)\,;
\end{equation}
\begin{equation}\label{f1infty}
-\infty < \displaystyle\liminf_{s\to +\infty} \frac{F(s)}{s^p}\leq \limsup_{s\to +\infty} \frac{F(s)}{s^p} = +\infty\,,
\end{equation}
where $p$ is the parameter given in \eqref{crescita} and $F$ is as in \eqref{F}\,.

As in the case of the oscillations near the origin, here we can construct a prototype for $f$ as follows
$$f(s)=\alpha s^{\alpha-1}\Big(1-\sin s^{\sigma}\Big)-\sigma s^{\alpha+\sigma-1} \cos s^{\sigma}-p\gamma s^{p-1}\,,$$
where $\alpha$, $\sigma$ and $\gamma$ are such that $\alpha>p$, $\sigma>0$ and $\gamma>0$.
Note that $f$ is continuous in $[0, +\infty)$ and $F$ is the following function
$$F(s)=\int_0^s f(t)\,dt=s^\alpha \Big(1-\sin s^{\sigma}\Big)-\gamma s^p,\,\,\, s>0\,.$$
Also in this case, direct calculations show that $f$ and $F$ satisfy assumptions \eqref{f2infty} and \eqref{f1infty}.

\medskip

In this setting the counterpart of Theorem~\ref{thorigin} can be stated as follows.

\begin{theorem}\label{thinfty}
Let $\Omega\subset \RR^N$, $N \geq 3$, be a bounded domain with smooth boundary, $\lambda\in \RR$ and let $A:\overline \Omega \times \RR^N\to \RR^N$ and $a:\overline\Omega\times \RR^N\to \RR$ be two functions satisfying \eqref{Acontinua}--\eqref{potenziale} and \eqref{acontinua}--\eqref{Aconvex}, respectively. Assume that $\beta \in L^{\infty}(\Omega)$ and $f \in C([0,+\infty);\RR)$ satisfies \eqref{f2infty}, \eqref{f1infty} and $f(0)=0$. If either
\begin{enumerate}
\item[$a)$] $q = p-1$, $\ell_\infty\in (0, +\infty)$ and $\lambda \beta(x) < \lambda_{\infty}$ a.e. $x\in \Omega$ for some $\lambda_{\infty}\in(0, \ell_\infty)$ or
\item[$b)$] $q = p-1$, $\ell_\infty=+\infty$ and $\lambda \in \RR$ is arbitrary or
\item[$c)$] $0<q<p-1$ and $\lambda \in \RR$ is arbitrary,
\end{enumerate}
then there exists a sequence $\{u_j\}_j$ in  $W_0^{1,p}(\Omega)$ of distinct weak solutions of problem~\eqref{problem} such that
\begin{equation} \label{1.3.0}
\lim_{j\to +\infty} \|u_j\|_{L^{\infty}(\Omega)}=+\infty\,.
\end{equation}

While, if $q>p-1$, then for every $k\in \NN$ there exists $\Lambda_k>0$ such that problem~\eqref{problem} has at least $k$ distinct weak solutions $u_1, \dots, u_k\in W^{1,p}_0(\Omega)$
such that
\begin{equation} \label{1.3.0bis}
\|u_j\|_{L^{\infty}(\Omega)}\geq j-1\,, \,\,\,\,\, j=1,\dots, k
\end{equation}
provided $|\lambda|<\Lambda_k$.
\end{theorem}

As in the case when there is an oscillation near the origin, here assumption~\eqref{f2infty} is used in order to prove the existence of solutions for problem~\eqref{problem}, while \eqref{f1infty} guarantees that these solutions are infinitely many, when $0<q\leq p-1$, and at least a finite number, if $q>p-1$.

In all the situations, that is, when there is an oscillation near zero or at infinity and for any value of $q$, the idea is to prove the existence of solutions for problem~\eqref{problem} using variational method. More precisely, we first consider an auxiliary problem and, under suitable assumptions on the data, we  prove the existence of solutions for this equation studying the associated energy functional and proving that this functional admits a minimum, using the direct methods of the calculus of variations (see Theorem~\ref{thauxiliary}).
Next, we  apply Theorem~\ref{thauxiliary} to problem~\eqref{problem}, in order to get Theorem~\ref{thorigin} and Theorem~\ref{thinfty}.

%Before ending this section, we would like to point out that the main results of the present paper (Theorems %\ref{thorigin} and \ref{thinfty}) generalize some recent contributions obtained in \cite[Theorems~1.1., 1.2, %1.3 and 1.4]{KM} for Laplacian problems, to the $p$-Laplacian case with $p>1$.


\section{Some comments on the assumptions}\label{sec:comments}
In this section we comment the assumptions on the data of problem~\eqref{problem} and we prove some preliminary results which will be useful in the sequel.

First of all, we would like to emphasize that the technical assumptions \eqref{Acontinua}--\eqref{Aconvex} appeared very recently in \cite{CPV}, where the authors studied elliptic equations involving operators in divergence form and proved the existence of at least three nontrivial weak solutions for their problem.

Now, we derive some relations involving $A$ and $a$. Since $a$ is the potential of $A$ and \eqref{condA} holds true, it is easily seen that for any $(x, \xi)\in \overline \Omega \times \RR^N$
\begin{equation}\label{aint}
a(x,\xi)=\int_0^1 \frac{d( a(x, t\xi))}{dt}\,dt=\int_0^1 A(x,t\xi)\cdot\xi\,dt\,,
\end{equation}
thanks to \eqref{condA}.

Furthermore, assumption~\eqref{crescita} implies
\begin{equation}\label{Anorma1}
\int_0^1 A(x,t\xi)\cdot \xi\,dt=\int_0^1 \frac 1 t A(x,t\xi)\cdot t\xi\,dt\geq \frac{\Gamma_1}{p}|\xi|^p
\end{equation}
and
\begin{equation}\label{Anorma2}
\int_0^1 A(x,t\xi)\cdot \xi\,dt\leq |\xi|\int_0^1 |A(x,t\xi)|\,dt\leq \frac{\Gamma_2}{p}|\xi|^p
\end{equation}
for any $(x,\xi) \in \Omega \times \RR^N$\,.

Hence, as a consequence of \eqref{aint}--\eqref{Anorma2}, we easily get that
\begin{equation} \label{box2}
\frac{\Gamma_1}{p}|\xi|^p \leq a(x,\xi)\leq \frac{\Gamma_2}{p} |\xi|^p,
\end{equation}
for every $(x,\xi)\in \Omega\times\RR^N$.

Thus, for every $u\in W^{1,p}_0(\Omega)$,
\begin{equation}\label{anorma}
\frac{\Gamma_1}{p}\|u\|_{W^{1,p}_0(\Omega)}^p\leq \int_\Omega a(x,\nabla u(x))dx\leq \frac{\Gamma_2}{p} \|u\|_{W^{1,p}_0(\Omega)}^p\,,
\end{equation}
that is
$$u\mapsto \int_\Omega a(x,\nabla u(x))dx$$
is a norm on $W^{1,p}_0(\Omega)$ which is equivalent to the usual one.

In the literature, when dealing with general second order operators in divergence form, the standard condition required on $a$ is the $p$--uniformly convexity, that is, that there exists a constant $K>0$ such that
\begin{equation}\label{pconvex}
a\left(x,\frac{\xi+\eta}{2}\right)\leq\frac{1}{2}\,a(x,\xi)
+\frac{1}{2}\,a(x,\eta)-K|\xi-\eta|^{p},
\end{equation}
for every $x\in\Omega$ and $\xi,\eta\in \RR^N$ (see, for instance, \cite{denapoli, MR} and references therein). We would like to note that condition~\eqref{Aconvex} is weaker that \eqref{pconvex}. Indeed, the function $a(x,\xi)=|\xi|^p/p$ satisfies \eqref{Aconvex} for any $p>1$, while verifies \eqref{pconvex} just when $p\geq 2$ (for this see \cite{denapoli}).

Before ending this section, we would like to discuss a property of the function $f$, which will be useful in the sequel. As a consequence of assumptions~\eqref{f2origin} and \eqref{f1origin} we have that
\begin{equation}\label{f0}
f(0)=0.
\end{equation}
Indeed, suppose that $f(0)=L\in \RR\setminus\{0\}$. Then, by the continuity of $f$ and \eqref{f2origin}
we would get
$$\lim_{s\to 0^+}\frac{f(s)}{s^{p-1}}=-\infty\,,$$
so that, by l'H\^opital's rule we would deduce that
$$\lim_{s\to 0^+}\frac{F(s)}{s^{p}}=\lim_{s\to 0^+}\frac{f(s)}{ps^{p-1}}=-\infty\,,$$
which contradicts \eqref{f1origin}. Hence, assertion~\eqref{f0} holds true.


\section{An auxiliary problem}\label{sec:auxiliary}
In this section we consider the problem
\begin{equation}\label{auxiliary}
\tag{$P_h^K$}
\left\{ \begin{array}{ll} -{\rm div}A(x,\nabla u) + K(x) |u|^{p-2}u = h(x, u)\,\,\,\, & \mbox{in} \, \Omega\\
 u=0 & \mbox{on} \, \partial \Omega.
 \end{array} \right.
\end{equation}

Here, we assume that $K:\Omega \to \RR$ is such that
\begin{equation}\label{K}
K \in L^{\infty}(\Omega)\,\,\, \mbox{with}\,\,\, \mathop {\rm ess\,inf}_{x\in \Omega} K(x)>0\,,
\end{equation}
while $h:\Omega \times [0,+\infty)\to \RR$ is a Carath\'eodory function satisfying the following conditions:
\begin{equation}\label{condh}
h(x, 0)=0\,\,\, \mbox{for a.e.}\, x \in \Omega\,;
\end{equation}
\begin{equation}\label{condhbis}
\mbox{there exists}\,\,\, M > 0\,\,\, \mbox{such that}\,\,\, |h(x, s)|\leq M\,\,\, \mbox{for a.e.}\, x\in \Omega\,\,\, \mbox{and for any}\, s \geq 0\,;
\end{equation}
\begin{equation}\label{hnegativo}
\begin{aligned}
& \mbox{there exist}\,\delta\, \mbox{and}\, \eta,\, \mbox{with}\,\, 0<\delta<\eta,\,\, \mbox{such that}\\
& h(x, s)\leq 0\,\,\, \mbox{for a.e.}\, x \in \Omega \,\,\, \mbox{and for any}\, s \in [\delta, \eta]\,.
\end{aligned}
\end{equation}
In the sequel we extend the function $h$ on the whole $\Omega \times \RR$ by taking $h(x, s)=0$ for a.e. $x\in \Omega$ and $s<0$.

The aim of this section is to prove the existence of a non-negative weak solution for problem~\eqref{auxiliary}, that is, a non-negative solution of the following problem:
\begin{equation}\label{auxiliaryweak}
\left\{\begin{aligned}
& \int_\Omega A(x,\nabla u(x))\nabla \varphi(x)\,dx+\int_\Omega K(x)|u(x)|^{p-2}u(x)\varphi(x)\,dx=\int_\Omega h(x, u(x))\varphi(x)\,dx\\
& \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \quad \mbox{for any}\,\, \varphi \in W^{1,\,p}_0(\Omega)\\
& u\in W^{1,\,p}_0(\Omega)\,.
\end{aligned}\right.
\end{equation}


Problem~\eqref{auxiliaryweak} has a variational nature and the energy functional~$\mathcal{E}_{K,\, h}: W_0^{1,p}(\Omega)\to \RR$ associated with it is defined as follows
\begin{equation}\label{EKh}
\mathcal{E}_{K,\, h}(u) = \int_\Omega a(x,\nabla u(x))\,dx+\frac{1}{p}\int_\Omega K(x)|u(x)|^p\,dx - \int_\Omega H(x,u(x))\,dx\,,
\end{equation}
where
\begin{equation}\label{H}
H(x, s):=\int_0^s h(x, t) dt \,\,\,\,\,\mbox{for any}\,\, s \in \RR\,.
\end{equation}

Due to \eqref{anorma}, hypotheses \eqref{K}--\eqref{hnegativo} and the embedding properties of the space $W^{1,p}_0(\Omega)$ into the Lebesgue spaces, it is easy to see that {$\mathcal{E}_{K,\, h}$} is well-defined. Moreover, standard arguments show that $\mathcal{E}_{K,\, h}$ is of class $C^1$ on $W_0^{1,p}(\Omega)$.

Hence, finding non-negative solutions of problem~\eqref{auxiliaryweak} means looking for non-negative critical points of the functional~$\mathcal{E}_{K,\,h}$. At this purpose, we introduce the set $W^\eta$ defined as follows
$$W^{\eta}:=\big\{u \in W_0^{1,p}(\Omega) \ : \ \|u\|_{L^{\infty}(\Omega)} \leq \eta \big\}\,,$$
where $\eta$ is the positive parameter given in \eqref{hnegativo}\,.

The main result of this section is given in the following theorem:
\begin{theorem}\label{thauxiliary}
Let $\Omega\subset \RR^N$, $N \geq 3$, be a bounded domain with smooth boundary, $\lambda\in \RR$ and let $A:\overline \Omega \times \RR^N\to \RR^N$ and $a:\overline\Omega\times \RR^N\to \RR$ be two functions satisfying \eqref{Acontinua}--\eqref{potenziale} and \eqref{acontinua}--\eqref{Aconvex}, respectively. Assume that $K:\Omega \to \RR$ is  a function verifying \eqref{K} and that $h:\Omega \times [0,+\infty)\to \RR$ is a Carath\'eodory function satisfying \eqref{condh}--\eqref{hnegativo}. Then,
\begin{enumerate}
\item[$i)$] the functional {$\mathcal{E}_{K,\, h}$} is bounded from below on $W^{\eta}$ and its infimum is attained at some $u_\eta \in W^{\eta}$\,;
\item[$ii)$] $u_\eta \in [0, \delta]$,\, where $\delta$ is the positive parameter given in \eqref{hnegativo};
\item[$iii)$] ${u}_\eta$ is a non-negative weak solution of problem~\eqref{auxiliary}.
\end{enumerate}
\end{theorem}
\begin{proof}
Let us start by proving assertion~$i)$\,. First of all, it is easy to see that the set $W^{\eta}$ is convex. Moreover, $W^\eta$ is closed in $W_0^{1,p}(\Omega)$. To see this, let $\{u_j\}_j$ be a sequence in $W^\eta$ such that $u_j\to u$ in $W_0^{1,p}(\Omega)$ as $j\to +\infty$.

We claim that $u\in W^\eta$. Of course $u\in W_0^{1,p}(\Omega)$. Furthermore, by assumption $\{u_j\}_j$ is bounded in $L^\infty(\Omega)$. Since $L^\infty(\Omega)$ is the dual space of $L^1(\Omega)$, which is a separable Banach space, then by \cite[Corollary~III.26]{brezis} we get that $u_j\to u$ in the weak$^*$ topology of $L^\infty(\Omega)$ as $j\to +\infty$. Hence, \cite[Proposition~III.12]{brezis} yields that, up to a subsequence, still denoted by $\{u_j\}_j$,
$$\liminf_{j\to +\infty}\|u_j\|_{L^\infty(\Omega)}\geq \|u\|_{L^\infty(\Omega)}\,.$$

As a consequence of this and taking into account that
$$\|u_j\|_{L^\infty(\Omega)}\leq \eta$$
for any $j\in \NN$, we get that
$$\|u\|_{L^\infty(\Omega)}\leq \eta\,,$$
that is $u\in W^\eta$, which proves the claim.

Thus, since $W^\eta$ is convex and closed in $W_0^{1,p}(\Omega)$, then it is weakly closed in $W_0^{1,p}(\Omega)$ by \cite[Theorem~III.7]{brezis}.

Now, let us consider the functional~$\mathcal{E}_{K,\,h}$\,. One can easily see that $\mathcal{E}_{K,\, h}$ is sequentially weakly lower semicontinuous (see \cite[Lemma 3.4.3]{Cola} for details). Moreover, note that \eqref{anorma}, \eqref{K}, \eqref{condhbis} and the definition of $H$ yield that the functional {$\mathcal{E}_{K,\, h}$} is bounded from below on $W^\eta$. Indeed, for any $u \in W^{\eta}$
$$\begin{aligned}
\mathcal{E}_{K,\, h}(u) & = \int_\Omega a(x,\nabla u(x))\,dx+\frac{1}{p}\int_\Omega K(x)|u(x)|^p\,dx - \int_\Omega H(x,u(x))\,dx\\
& \geq \frac{\Gamma_1}{p}\|u\|_{W^{1,p}_0(\Omega)}^p- \int_\Omega H(x,u(x))\,dx\\
&\geq - \int_\Omega H(x,u(x))\,dx\\
& \geq -M\int_\Omega |u(x)|\,dx\\
&\geq -\eta{M}\mathcal{L}(\Omega)\,,
\end{aligned}$$
where $\mathcal L(\Omega)$ denotes the Lebesgue measure of $\Omega$.

Let us denote by $\alpha_\eta$ the infimum of $\mathcal{E}_{K,\,h}$ on $W^\eta$, that is,
\begin{equation}\label{alfan}
\alpha_\eta:=\displaystyle\inf_{u\in W^\eta}\mathcal{E}_{K,\, h}(u)>-\infty\,.
\end{equation}
It is easily seen that for every $k \in \NN$, there exists $u_k\in W^\eta$ such that
\begin{equation}\label{inf1}
\alpha_\eta\leq \mathcal{E}_{K,\, h}(u_k)\leq\alpha_\eta+\frac{1}{k}\,.
\end{equation}
Also, since $u_k\in W^\eta$ and thanks to \eqref{condhbis}, we get
$$\begin{aligned}
\displaystyle \int_\Omega a(x,\nabla u_k(x))dx+\frac{1}{p}\int_\Omega K(x)|u_k(x)|^p dx &= \int_\Omega H(x,u_k(x)) dx+ \mathcal{E}_{K,\, h}(u_k)\\
&\leq \eta{M}\mathcal{L}(\Omega)+\mathcal{E}_{K,\, h}(u_k)\\
&\leq \eta{M}\mathcal{L}(\Omega)+\alpha_\eta+\frac{1}{k}\\
&\leq \eta{M}\mathcal{L}(\Omega)+\alpha_\eta+1,
\end{aligned}$$
for every $k\in\NN$. Thus, by \eqref{anorma} and \eqref{K}
\begin{equation}\label{vkconv}
\|u_k\|_{W_0^{1,p}(\Omega)}^p\leq \frac{p}{\Gamma_1}\Big(\eta{M}\mathcal{L}(\Omega)+\alpha_\eta+1\Big),
\end{equation}
for every $k\in\NN$. Then, the sequence $\{u_k\}_k$ is bounded in $W_0^{1,p}(\Omega)$ and so, up to a subsequence, still denoted by $\{u_k\}_k$,
\begin{equation}\label{vweakly222}
u_k\to u_\eta\,\,\,\, \mbox{weakly in}\,\, W_0^{1,p}(\Omega)
\end{equation}
as $k\to +\infty$ for some $u_\eta\in W_0^{1,p}(\Omega)$.

Now, let us show that $u_\eta$ is the minimum of $\mathcal{E}_{K,\,h}$ we are looking for. At this purpose, first of all note that $u_\eta\in W^{\eta}$, since $W^\eta$ is weakly closed in $W_0^{1,p}(\Omega)$.
Thus, by \eqref{alfan}
\begin{equation}\label{disE1}
\mathcal{E}_{K,\,h}(u_\eta)\geq \alpha_\eta\,.
\end{equation}

On the other hand, thanks to the sequential weak lower semicontinuity of $\mathcal{E}_{K,\, h}$, \eqref{inf1} and \eqref{vweakly222}, we obtain that
$$\alpha_\eta\geq \liminf_{k\to +\infty}\mathcal{E}_{K,\,h}(u_k)\geq \mathcal{E}_{K,\,h}(u_\eta)\,.$$
By this and \eqref{disE1} we obtain that
$$\mathcal{E}_{K,\, h}(u_\eta)=\alpha_\eta\,.$$
This and \eqref{alfan} conclude the proof of statement~$i)$\,.

\medskip

Now, let us prove $ii)$. At this purpose, let $\delta$ be as in assumption~\eqref{hnegativo} and let $B$ the following set
$$B := \{x\in \Omega \ : \ {u}_\eta(x) \notin [0, \delta]\}\,.$$
We argue by contradiction and we suppose that $\mathcal{L}(B) > 0$.

Let $\gamma : \RR \rightarrow \RR$ be the function given by
$$\gamma(s) := \min\{s_+, \delta\},$$
where $s_+ = \max\{s, 0\}$. Also, set $w := \gamma \circ {u}_\eta$, that is
$$w(x)=
\left\{ \begin{array}{lrl}
\delta & \mbox{if}  & u_\eta(x)> \delta \\
u_\eta(x)     & \mbox{if}  & 0\leq u_\eta(x) \leq \delta \\
0     & \mbox{if}  & u_\eta(x)< 0,
\end{array} \right.
$$
for a.e. $x\in \Omega$.

Since $\gamma$ is a Lipschitz function and $\gamma(0) = 0$, the theorem of Marcus-Mizel (see [9]) shows that $w \in W_0^{1,p}(\Omega)$. Moreover, $0 \leqslant w(x) \leqslant \delta$ for a.e. $\Omega$. Consequently, $w \in W^{\eta}$, being $\delta<\eta$, by assumption~\eqref{hnegativo}.

We introduce the sets
$$B_1 := \{x \in \Omega \ : \ {u}_\eta(x) < 0\}$$ and $$B_2 := \{x \in \Omega \ : \ u_\eta(x) > \delta\}.$$
Thus, $B = B_1 \cup B_2$, and we have that
$w(x)={u}_\eta(x)$ for a.e. $x\in \Omega \setminus B, w(x) = 0$ for a.e. $x\in B_1$, and $w(x) = \delta$ for a.e. $x\in B_2$.

As a consequence of this and of \eqref{condA} we get
$$\displaystyle \left(\displaystyle \int_{\Omega}a(x,\nabla w(x))dx- \displaystyle \int_{\Omega}a(x,\nabla {u}_\eta(x))dx\right)=\displaystyle -\int_{B}a(x,\nabla {u}_\eta(x))dx,$$
from which it follows that
\begin{equation}\label{eq222}
\begin{aligned}
\mathcal{E}_{K,\, h}(w)-\mathcal{E}_{K,\, h}({u}_\eta) & = \left(\displaystyle \int_{\Omega}a(x,\nabla w(x))dx- \int_{\Omega}a(x,\nabla {u}_\eta(x))dx\right)\\
& \qquad \qquad + \frac{1}{p}\int_{\Omega} K(x)\Big(|w(x)|^p-|{u}_\eta(x)|^p\Big)dx\\
& \qquad \qquad  - \int_{\Omega} \Big(H(x,w(x))-H(x,{u}_\eta(x))\Big)dx\\
& = -\int_{B}a(x,\nabla {u}_\eta(x))dx +\frac{1}{p}\int_B K(x)\Big(|w(x)|^p-|{u}_\eta(x)|^p\Big)dx\\
& \qquad \qquad - \int_B \Big(H(x,w(x))-H(x,{u}_\eta(x))\Big)dx\,.
\end{aligned}
\end{equation}
Since ${\displaystyle \mathop {\rm ess\,inf}_{x\in \Omega} K(x)>0}$ by \eqref{K}, one has
\begin{equation}\label{eqK}
\begin{aligned}
\int_B K(x)\Big(|w(x)|^p-|{u}_\eta(x)|^p\Big)dx & =-\int_{B_1} K(x)|{u}_\eta(x)|^pdx\\
& \qquad \qquad +\int_{B_2} K(x)\Big(\delta^p-|{u}_\eta(x)|^p\Big)dx \leqslant 0.
\end{aligned}
\end{equation}
Moreover, due to the fact that $h(x,s)=0$ for a.e. $x\in \Omega$ and all $s\leq 0$, then
\begin{equation}\label{HB1}
\int_{B_1} \Big(H(x,w(x))-H(x,{u}_\eta(x))\Big)dx=0\,,
\end{equation}
while, by the mean value theorem, for a.e. $x\in B_2$, there exists $\theta(x)\in [\delta,{u}_\eta(x)] \subseteq [\delta,\eta]$ such that
$$H(x,w(x))-H(x,{u}_\eta(x))=H(x,\delta)-H(x,{u}_\eta(x)) =h(x,\theta(x))(\delta-{u}_\eta(x)).$$
Thus, taking into account \eqref{hnegativo} and the definition of $B_2$, one has
\begin{equation}\label{HB2}
\int_{B_2} \Big(H(x,w(x))-H(x,{u}_\eta(x))\Big)dx=\int_{B_2} h(x,\theta(x))(\delta-{u}_\eta(x))dx\geq 0\,.
\end{equation}

Hence, by \eqref{HB1} and \eqref{HB2}, we get that
\begin{equation}\label{HB}
\int_B \Big(H(x,w(x))-H(x,{u}_\eta(x))\Big)dx\geq 0\,.
\end{equation}
As a consequence of \eqref{eq222}, \eqref{eqK}, \eqref{HB} and taking into account \eqref{anorma}, we get
\begin{equation}\label{Enegativo}
\mathcal{E}_{K,\, h}(w)-\mathcal{E}_{K,\, h}(u_\eta)\leq 0\,.
\end{equation}

On the other hand, since $w \in W^{\eta}$, it is easy to see that $\mathcal{E}_{K,\, h}(w)\geqslant \mathcal{E}_{K,\, h}({u}_\eta)$. By this and \eqref{Enegativo} we get that
\begin{equation}\label{Ezero}
\mathcal{E}_{K,\, h}(w)=\mathcal{E}_{K,\, h}({u}_\eta)\,.
\end{equation}
Since \eqref{Ezero} holds true and all the integrals in the right hand-side of \eqref{eq222} are non-negative, it is easy to see that every integral term in \eqref{eq222} should be zero.
In particular,
$$\int_{B_1} K(x)|{u}_\eta(x)|^p = \int_{B_2} K(x)\Big(|{u}_\eta(x)|^p-\delta^p\Big)dx=0\,.$$
Due to the definition of $B_1$ and $B_2$ and to \eqref{K}, we necessarily have $\mathcal{L}(B_1) = \mathcal{L}(B_2)= 0$, that is $\mathcal{L}(B)=0$, contradicting our assumption.
Thus, assertion~$ii)$ is proved.

\medskip

Finally, let us show $iii)$. For this let us fix $\varphi\in C_0^{\infty}(\Omega)$ and let
$$\varepsilon_0 := \frac{\eta - \delta}{\|\varphi\|_{L^{\infty}(\Omega)}+ 1} > 0\,,$$
where $\delta$ and $\eta$ are given as in \eqref{hnegativo}. Moreover, let $E : [-\varepsilon_0, \varepsilon_0]\to \RR$ be the function defined as $$E(\varepsilon)=\mathcal{E}_{K,\, h}({u}_\eta + \varepsilon \varphi)\,.$$
First of all, note that, since $ii)$ holds true, for any $\varepsilon\in [-\varepsilon_0, \varepsilon_0]$ we have
$$\begin{aligned}
|{u}_\eta(x)+\varepsilon \varphi(x)|& \leq |{u}_\eta(x)|+\varepsilon|\varphi(x)|\\
& \leq  {u}_\eta(x)+\frac{\eta-\delta}{\|\varphi\|_{L^{\infty}(\Omega)}+1}
\|{\varphi}\|_{L^{\infty}(\Omega)}\\
& \leq \delta +\eta-\delta =\eta\,,
\end{aligned}$$
for a.e. $x\in \Omega$. So, ${u}_\eta+\varepsilon \varphi\in W^\eta$\,.

Consequently, due to $i)$, one has $E(\varepsilon) \geq E(0)$ for every $\varepsilon\in [-\varepsilon_0, \varepsilon_0]$\,, that is $0$ is an interior minimum point for $E$. Then, since $E$ is differentiable at $0$, it is easy to see that $E'(0) = 0$ and so also $\langle\mathcal{E}_{K,\, h}'({u}_\eta), \varphi\rangle = 0$. Taking into account that $\varphi\in C_0^{\infty}(\Omega)$ is arbitrary and using the definition of $\mathcal{E}_{K,\,h}$, we obtain that $u_\eta$ is a weak solution of problem~\eqref{auxiliary} (that is a solution of \eqref{auxiliaryweak}). Of course, $u_\eta$ is non-negative in $\Omega$ thanks to $ii)$ and this ends the proof of $iii)$.
\end{proof}

We would like to note that, as a consequence of \eqref{crescita}, it is easily seen that $A(x,0)=0$ for any $x\in \Omega$. Hence, since also $h(x,0)=0$ a.e. $x\in \Omega$ by \eqref{condh}, the function $u\equiv 0$ is a weak solution of problem~\eqref{auxiliary}.
Theorem~\ref{thauxiliary} does not guarantee that the solution $u_\eta$ of problem~\eqref{auxiliary} is not the trivial one. In spite of this, by Theorem~\ref{thauxiliary} we will derive the existence of non-trivial solutions for the original problem~\eqref{problem}, provided the nonlinear term~$f$ is chosen appropriately.

\medskip

We conclude this section by constructing a special function which will be useful in the proof of our main theorems.
In the sequel, let $x_0\in \Omega$ and $r>0$ be such that $B(x_0, r) \subset \Omega$. For any $s >0$ we define the function~$z_s$ as follows:
\begin{equation} \label{2.1.1}
z_s(x):=\left\{
\begin{array}{ll}
0 & \mbox{ if $x \in \Omega \setminus B(x_0,r)$} \\
\displaystyle\frac{2s}{r} \left(r- |x-x_0| \right)
& \mbox{ if $x \in B(x_0,r) \setminus B(x_0,{r}/{2})$} \\
s & \mbox{ if $x \in B(x_0,{r}/{2})$}\,.
\end{array}
\right.
\end{equation}
It is clear that $z_s\geq 0$ in $\Omega$ and $z_s \in W_0^{1,p}(\Omega)$. Moreover, $\|z_s\|_{L^{\infty}(\Omega)}=s$ and
\begin{equation}\label{2.1.2}
\|z_s\|^p_{W^{1,p}_0(\Omega)}=\int_\Omega |\nabla z_s(x)|^p\,dx \leq \frac{2^ps^p\omega_N r^N}{r^p}\equiv C(r,p,N)s^p,
\end{equation}
where $C(r,p,N)$ is a positive constant and $\omega_N$ is the volume of the unit ball in $\RR^N$.

We also introduce the truncation function $\tau_{\eta} : [0,+\infty) \to \RR$ defined as
\begin{equation}\label{tau}
\tau_{\eta}(s) := \min\{\eta, s\}
\end{equation}
for any $s \geq 0$\,, where $\eta$ is the positive constant given in assumption~\eqref{hnegativo}. Note that $\tau_\eta$ is a continuous function in $[0,+\infty)$\,.

\section{Oscillation near the origin}\label{sec:origin}
In this section we study problem~\eqref{problem} in the case when the nonlinear term~$f$ oscillates near the origin.

In order to prove Theorem~\ref{thorigin} we first give an auxiliary result obtained as a consequence of Theorem~\ref{thauxiliary}. Precisely, we prove the existence of infinitely many solutions for problem~\eqref{auxiliary} under the following assumptions on the function~$h$:
\begin{equation}\label{h1origin}
\mbox{there exists}\,\,\, \bar s > 0\,\,\, \mbox{such that}\\
\sup_{s\in [0,\bar s]} |h(\cdot, s)|\in L^{\infty}(\Omega)\,;
\end{equation}
\begin{equation}\label{h3origin}
\begin{aligned}
& \qquad  \mbox{there exist two sequences}\,\,\, \{\delta_j\}_j\,\,\mbox{and}\,\, \{\eta_j\}_j\,\,\, \mbox{with}\,\,\, 0 < \eta_{j+1} < \delta_j < \eta_j\,\, \mbox{and}\\
& \displaystyle\lim_{j\to +\infty} \eta_j = 0\,\, \mbox{such that}\,\,\, h(x, s) \leq 0\,\,\, \mbox{for a.e.}\, x\in \Omega\,\,\, \mbox{and for every}\, s\in [\delta_j,\eta_j],\,\, j\in \NN\,;
\end{aligned}
\end{equation}
\begin{equation}\label{h2origin}
-\infty< \displaystyle\liminf_{s\to 0^+}\displaystyle \frac{H(x,s)}{s^p}\leq\displaystyle\limsup_{s\to 0^+} \displaystyle\frac{H(x,s)}{s^p}=+\infty\,\, \mbox{uniformly for a.e.}\,\,\, x\in \Omega\,,
\end{equation}
where $H$ is the function given in \eqref{H}\,.

In this setting our result for problem~\eqref{auxiliary} is the following:

\begin{theorem}\label{thauxorigin}
Let $\Omega\subset \RR^N$, $N \geq 3$, be a bounded domain with smooth boundary, $\lambda\in \RR$ and let $A:\overline \Omega \times \RR^N\to \RR^N$ and $a:\overline\Omega\times \RR^N\to \RR$ be two functions satisfying \eqref{Acontinua}--\eqref{potenziale} and \eqref{acontinua}--\eqref{Aconvex}, respectively. Moreover, assume that $K:\Omega \to \RR$ satisfies \eqref{K} and $h:\Omega \times [0,+\infty)\to \RR$ is a Carath\'eodory function verifying \eqref{condh} and \eqref{h1origin}--\eqref{h2origin}.

Then, there exists a sequence $\{u_j\}_j \subset W_0^{1,p}(\Omega)$ of distinct non-trivial non-negative weak solutions of problem \eqref{auxiliary} such that
\begin{equation} \label{3.1.0}
\lim_{j\to +\infty} \|u_j\|_{W^{1,p}_0(\Omega)}= \lim_{j\to +\infty} \|u_j\|_{L^{\infty}(\Omega)} = 0.
\end{equation}
\end{theorem}
\begin{proof}
Since $\eta_j\to 0$ as $j\to +\infty$, by \eqref{h3origin}, without loss of generality, we may assume that
\begin{equation}\label{AA}
\delta_j<\eta_j<\bar s
\end{equation}
for $j$ sufficiently large, where $\bar s > 0$ comes from \eqref{h1origin}.

For every $j\in \NN$, let $h_j:\Omega \times [0,+\infty) \to \RR$ be the function defined by
\begin{equation} \label{3.1.1}
h_j(x,s)=h(x,\tau_{\eta_j}(s)),
\end{equation}
and
$$H_j(x, s) :=\displaystyle \int^s_0 h_j(x, t) dt$$
for a.e. $x\in \Omega$ and $s\geq 0$, where $\tau_{\eta_j}$ is the function defined in \eqref{tau} with $\eta=\eta_j$. Also, in what follows for any $j\in \NN$ we denote by
\begin{equation}\label{funzionaleE}
\mathcal{E}_j:= \mathcal E_{K,\,h_j}\,,
\end{equation}
where $\mathcal E_{K,\,h_j}$ is the functional given in \eqref{EKh}, with $h=h_j$. Note that $\mathcal E_j$ is the energy functional associated with problem~$(P^K_{h_j})$, that is \eqref{auxiliary} with $h=h_j$.

The function $h_j$ verifies all the assumptions of Theorem~\ref{thauxiliary} for $j\in \NN$ large enough. Indeed, due to the regularity of $h$, the continuity of $\tau_\eta$ and \eqref{condh}, the function~$h_j$ is Carath\'eodory and such that $h_j(x,0)=0$ a.e. $x\in \Omega$. Moreover, by \eqref{h1origin}, \eqref{AA} and \eqref{3.1.1}, $h_j$ satisfies \eqref{condhbis}. Finally, condition~\eqref{hnegativo} comes from \eqref{h3origin}.

Hence, as a consequence of Theorem~\ref{thauxiliary}, for $j$ sufficiently large there exists $u_j \in W^{\eta_j}$ such that
\begin{equation} \label{3.1.2}
\min_{u\in W^{\eta_j}}\mathcal E_j(u)=\mathcal E_j(u_j)
\end{equation}
\begin{equation} \label{3.1.3}
u_j (x)\in [0,\delta_j] {\rm \,\,\,\,\, for \,\, a.e.} \,\, x\in \Omega,
\end{equation}
and
\begin{equation} \label{3.1.4}
u_j\,\,\, \mbox{is a non-negative weak solution of} \,\, (P^K_{h_j}).
\end{equation}

By the definition of $\tau_\eta$, \eqref{3.1.1} and the fact that $u_j(x)\leq \delta_j<\eta_j$ a.e. $x\in \Omega$, then
$$h_j(x,u_j(x))=h(x, \tau_{\eta_j}(u_j(x))=h(x,u_j(x))$$
a.e. $x\in \Omega$. Thus, by this and \eqref{3.1.4}, $u_j$ is a non-negative weak solution not only for $(P^K_{h_j})$ but also for problem~\eqref{auxiliary}.

To conclude the proof of Theorem~\ref{thauxorigin}, we have to prove that there are infinitely many distinct elements in the sequence $\{u_j\}_j$. In order to see this, we first claim that
\begin{equation}\label{claim1}
\mathcal{E}_j(u_j)<0\,\, \mbox{for}\,\,\, j \in \NN\,\,\, \mbox{large enough}.
\end{equation}

Assumption~\eqref{h2origin} implies the existence of some $\ell > 0$ and $\zeta\in (0,\eta_1)$ such that
\begin{equation} \label{3.1.7}
\mathop {\rm ess\,inf}_{x\in \Omega} H(x, s) \geq -\, \ell s^p\,\,\, \mbox{for all} \,\,\, s\in (0,\zeta)
\end{equation}
and that
there is a sequence $\{s_j\}_j$ such that $0<s_j \to 0$ as $j\to +\infty$ (here we use the definition of $H$, to take $s_j>0$) such that
\begin{equation} \label{3.1.9lim}
\lim_{j\to +\infty}\frac{ \mathop {\rm ess\,inf}_{x\in \Omega} H(x, s_j)}{s_j^p}=+\infty\,, \end{equation}
namely, for any $L>0$
\begin{equation}\label{3.1.9}
\mathop {\rm ess\,inf}_{x\in \Omega} H(x, s_j)> L s_j^p
\end{equation}
for $j\in \NN$ large enough.

Since $\delta_j\searrow 0$ as $j\to +\infty$, we can choose a subsequence of $\{\delta_j\}_j$, still denoted by $\{\delta_j\}_j$, such that
\begin{equation}\label{BB}
s_j\leq \delta_j
\end{equation}
for all $j\in \NN$.

Now, let us fix $j\in \NN$ sufficiently large and let
$$z_j:=z_{s_j}\in W_0^{1,p}(\Omega)$$
be the function defined as in \eqref{2.1.1} with $s=s_j$. Then, $z_j\in W^{1,p}_0(\Omega)$ and $\|z_j\|_{L^\infty(\Omega)}=s_j\leq\delta_j< \eta_j$ by \eqref{BB} and \eqref{h3origin}. Hence, $z_j\in W^{\eta_j}$ and $0\leq z_j(x)\leq s_j\leq \delta_j<\eta_j$ a.e. $x\in \Omega$. This yields that for a.e. $x\in \Omega$
$$\int_0^{z_j(x)}h_j(x,t)dt=\int_0^{z_j(x)}h(x,\tau_{\eta_j}(t))dt=\int_0^{z_j(x)}h(x,t)dt\,.$$
 By this and taking into account \eqref{anorma}, \eqref{K} and \eqref{2.1.2}, for $j$ sufficiently large one has:
\begin{equation}\label{Ejzj}
\begin{aligned}
\mathcal{E}_j(z_j) & = \displaystyle \displaystyle \int_\Omega a(x,\nabla z_j(x))dx+\frac{1}{p}\int_\Omega K(x)|z_j(x)|^pdx - \int_\Omega H_j(x,z_j(x)) dx\\
& = \displaystyle \displaystyle \int_\Omega a(x,\nabla z_j(x))dx+\frac{1}{p}\int_\Omega K(x)|z_j(x)|^pdx - \int_\Omega H(x,z_j(x)) dx\\
& \leq \displaystyle C(r,p,N)\frac{\Gamma_2}{p} s_j^p + \frac{1}{p}\int_\Omega K(x)|z_j(x)|^pdx\\
& \qquad \qquad \displaystyle - \int_{B(x_0,r/2)} H(x, s_j)dx -
\int_{B(x_0,r) \setminus B(x_0,r/2)} H(x, z_j(x))dx\\
& \displaystyle \leq \Big(C(r,p,N)\frac{\Gamma_2}{p} +  \|K\|_{L^{\infty}(\Omega)} \frac{\mathcal{L}(\Omega)}{p} - L(r/2)^N \omega_N + \ell \mathcal{L}(\Omega) \Big) s_j^p\,,
\end{aligned}
\end{equation}
thanks to \eqref{3.1.7}, \eqref{3.1.9} and using the fact that $z_j(x)<\eta_j< \eta_1$ (being $\{\eta_j\}_j$ decreasing by \eqref{h3origin}). Here $\omega_N$ denotes the volume of the unit ball in $\RR^N$\,.
Choosing $L>0$ large enough so that
$$L(r/2)^N \omega_N>C(r,p,N)\frac{\Gamma_2}{p} +  \|K\|_{L^{\infty}(\Omega)} \frac{\mathcal{L}(\Omega)}{p} + \ell \mathcal{L}(\Omega)\,,$$
we get that, for $j$ large enough
$$\mathcal{E}_j(z_j)<0\,.$$
Consequently, using also \eqref{3.1.2}, we obtain that, if $j$ is sufficiently large
\begin{equation} \label{3.1.10}
\begin{aligned}
& \mathcal{E}_j(u_j)= \displaystyle \min_{u\in W^{\eta_j}(u)} \mathcal{E}_j \leq \mathcal{E}_j (z_j)<0,
\end{aligned}
\end{equation}
which proves \eqref{claim1}. Also, this guarantees that $u_j\not \equiv 0$ in $\Omega$, being $\mathcal E_j(0)=0$\,.

Now, we claim that
\begin{equation}\label{claim2}
\lim_{j\to +\infty} \mathcal E_j(u_j)=0\,.
\end{equation}
At this purpose, note that for $j\in \NN$ sufficiently large, by using the definition of $H_j$, \eqref{h1origin}, \eqref{h3origin}, \eqref{AA}, \eqref{3.1.1} and \eqref{3.1.3}, we have
\begin{equation}\label{claim2proof}
\begin{aligned}
\mathcal{E}_j(u_j) & \geq \displaystyle - \int_\Omega H_j(x,u_j (x))dx \\
& =-\int_\Omega \int_0^{u_j(x)}h(x,s)\,ds\\
& \geq -\int_\Omega \sup_{s\in [0, \bar s]}|h(x,s)|\,u_j(x)dx\\
& \geq - \mathcal{L}(\Omega)\Big\| \sup_{s\in [0,\bar s]} |h(\cdot, s)\Big\|_{L^\infty(\Omega)} \delta_j\,.
\end{aligned}
\end{equation}

Since$\displaystyle\lim_{j\to +\infty} \delta_j = 0$ by \eqref{h3origin}, the above inequality and \eqref{3.1.10} leads to \eqref{claim2} and so the claim is proved.

Combining \eqref{claim1} and \eqref{claim2}, we deduce that the sequence $\{u_j\}_j$ contains infinitely many distinct elements, that is problem~\eqref{auxiliary} has infinitely many distinct weak solutions.

Finally, it remains to prove relation \eqref{3.1.0}. Since $\|u_j\|_{L^{\infty}(\Omega)} \leq \delta_j$ for $j\in \NN$ sufficiently large by \eqref{3.1.3}, and $\displaystyle \lim_{j\to +\infty} \delta_j = 0$ (see \eqref{h3origin}), we easily get that $\|u_j\|_{L^\infty(\Omega)}\to 0$ as $j\to +\infty$\,.

For the latter limit, observe that by \eqref{K}, we have
$$\begin{aligned}
\int_\Omega a(x,\nabla u_j(x))dx & \leq \displaystyle \int_\Omega a(x,\nabla u_j(x))dx  + \frac{1}{p} \int_\Omega K(x)|u_j(x)|^pdx\\
& <\displaystyle \int_\Omega H_j(x,u_j (x))dx\\
& =\displaystyle \int_\Omega H(x,u_j (x))dx\\
& \leq  \mathcal{L}(\Omega)\Big\| \sup_{s\in [0,\bar s]} |h(\cdot, s)|\Big\|_{L^\infty(\Omega)} \delta_j\,,
\end{aligned}$$
thanks to \eqref{h1origin}, \eqref{3.1.3} and \eqref{claim1}.

Thus, by \eqref{anorma} and \eqref{h3origin} it is easy to see that
$$
\lim_{j\to +\infty} \|u_j\|_{W^1_0(\Omega)}^p\leq \frac{p}{\Gamma_1}\lim_{j\to +\infty} \displaystyle \int_\Omega a(x,\nabla u_j(x))dx= 0\,,$$
which concludes the proof of Theorem~\ref{thauxorigin}\,.
\end{proof}

Now, we are ready to prove Theorem~\ref{thorigin}. The strategy will consists in applying Theorem~\ref{thauxiliary} and Theorem~\ref{thauxorigin} with a suitable choice of the functions $K$ and $h$.

\subsection{Proof of Theorem~\ref{thorigin}}\label{subsec:prooforigin}
First of all, we show that, under suitable assumptions, problem~\eqref{problem} has infinitely many distinct weak solutions, provided $q\geq p-1$.
Let us consider separately the case when $q=p-1$ and the one when $q>p-1$: in both the situations the strategy will consist in using Theorem~\ref{thauxorigin}.

Let us start proving assertion~$a)$. In this setting we suppose that $q = p-1$, $\ell_0\in (0, +\infty)$ and $\lambda \in \RR$ is such that $\lambda \beta(x) < \lambda_0$ a.e. $x\in \Omega$ for some $\lambda_0\in(0,\ell_0)$. Let us choose $\tilde{\lambda}_0 \in (\lambda_0, \ell_0)$ and let
\begin{equation} \label{1.1.1}
K(x) := \tilde{\lambda}_0 - \lambda \beta(x)\,\, \,\,\,{\rm and} \,\,\,\,\, h(x, s) := \tilde{\lambda}_0 s^{p-1} + f (s),
\end{equation}
a.e. $x\in \Omega$ and $s\geq 0$.

Now, we show that the functions~$K$ and $h$ given in \eqref{1.1.1} satisfy all the assumptions of Theorem~\ref{thauxorigin}. First of all, note that $K\in L^\infty(\Omega)$ and
$$\mathop {\rm ess\,inf}_{x\in \Omega} K(x)\geq \tilde{\lambda}_0 - \lambda_0 > 0\,,$$ thanks to the fact that $\beta\in L^\infty(\Omega)$. Hence, \eqref{K} is satisfied.

Moreover, using the regularity of $f$, it is easy to see that $h$ is a continuous function in $\Omega\times [0, +\infty)$ and $h(x, 0)=0$ for any $x\in \Omega$, since $f(0)=0$, due to \eqref{f0}. Thus, $h$ verifies assumption~\eqref{condh}. Also, the continuity of $s\mapsto h(\cdot, s)$ and the Weierstrass Theorem yield \eqref{h1origin}. Furthermore, since for any $x\in \Omega$ and $s>0$ $$\frac{H(x, s)}{s^p} = \frac{\tilde{\lambda}_0}{p} + \frac{F(s)}{s^p}\,,$$ hypothesis \eqref{f1origin} immediately implies \eqref{h2origin}.

It remains to show that $h$ satisfies \eqref{h3origin}. At this purpose, note that, by \eqref{f2origin}, we get that there exists a sequence $\{s_j\}_j$ converging to 0 such that
\begin{equation}\label{ell222}
\frac{f (s_j)}{s_j^{p-1}} \to -\ell_0
\end{equation}
as $j\to +\infty$\,. Now, since $\tilde \lambda_0<\ell_0$ by assumption, there exists $\overline \varepsilon>0$ such that $\tilde \lambda_0+\overline \varepsilon<\ell_0$. By this and \eqref{ell222} we get that, for $j$ large enough, say $j\geq j^*\in \NN$,
\begin{equation}\label{formulareplaced}
\frac{f (s_j)}{s_j^{p-1}} < -\tilde{\lambda}_0\,.
\end{equation}
Consequently, by using the continuity of $f$, there exists a neighborhood of $s_j$, say $(\delta_j, \eta_j)$ such that $$h(x,s)=\tilde{\lambda}_0 s^{p-1} + f (s) \leq 0,$$ for any $x\in \Omega$ and all $s\in [\delta_j,\eta_j]$ and $j\geq j^*$. Therefore, \eqref{h3origin} holds too.

Now, we can apply Theorem~\ref{thauxorigin} to problem~\eqref{auxiliary} with $K$ and $h$ given in \eqref{1.1.1}. As a consequence of this, we get the existence of infinitely many distinct non-trivial non-negative solutions $\{u_j\}_j$ for problem~\eqref{auxiliary}, satisfying condition~\eqref{1.1.0}
Due to the choice of $K$ and $h$ in \eqref{1.1.1} and taking into account that $q=p-1$, it is easy to see that $u_j$ is a weak solution of problem~\eqref{problem} and this ends the proof of Theorem~\ref{thorigin} in the case $q=p-1$.

\bigskip

Now, let us consider assertion~$b)$. At this purpose, let $q=p-1$, $\ell_0=+\infty$ and $\lambda \in \RR$. In this case we choose $\tilde{\lambda}_0 \in (\lambda_0, \ell_0)$ and
\begin{equation} \label{1.1.1infty}
K(x) := \tilde{\lambda}_0 \,\, \,\,\,{\rm and} \,\,\,\,\, h(x, s) := \Big(\lambda \beta(x)+\tilde{\lambda}_0\Big) s^{p-1} + f (s),
\end{equation}
a.e. $x\in \Omega$ and $s\geq 0$. In this setting we can argue exactly as in the proof of assertion~$a)$, just replacing formula~\eqref{formulareplaced} with the following one
\begin{equation}\label{formulareplaced2}
\frac{f (s_j)}{s_j^{p-1}} < -\Big(|\lambda|\|\beta\|_{L^\infty(\Omega)}+\tilde{\lambda}_0\Big)\,.
\end{equation}
for $j$ large enough, and taking into account that
$$h(x,s)=\Big(\lambda \beta(x)+\tilde{\lambda}_0\Big) s^{p-1} + f (s)\leq \Big(|\lambda|\|\beta\|_{L^\infty(\Omega)}+\tilde{\lambda}_0\Big)s^{p-1} + f (s)\,.$$

\bigskip

Now, let us prove assertion~$c)$. At this purpose, let $q>p-1$ and $\lambda \in \RR$. Let $\tilde\lambda_0\in (0,\ell_0)$ and
\begin{equation} \label{1.1.2}
K(x) := \tilde\lambda_0 \,\,\,{\rm and} \,\,\, h(x, s) := \lambda \beta(x) s^q + \tilde\lambda_0 s^{p-1} + f (s),
\end{equation}
for a.e. $x\in\Omega$ and $s\geq 0$.
Also in this setting our aim is to prove that $K$ and $h$ given in \eqref{1.1.2} satisfy the conditions required by Theorem~\ref{thauxorigin}.

Clearly, \eqref{K} and \eqref{condh} are trivially satisfied, also thanks to \eqref{f2origin}. Moreover, since $\beta\in L^{\infty}(\Omega)$, the continuity of $s\mapsto h(\cdot, s)$ and the Weierstrass Theorem yield that \eqref{h1origin} holds true.
Moreover, for a.e. $x\in \Omega$ and $s> 0$ we have
$$\frac{H(x, s)}{s^p} = \lambda  \frac{\beta(x)}{q+1}s^{q-p+1}+\frac{\tilde\lambda_0}{p}+\frac{F(s)}{s^p}\,,$$
so that hypothesis \eqref{f1origin} and the fact that $q>p-1$ imply \eqref{h2origin}.

Finally, note that for a.e $x\in \Omega$ and any $s\geq 0$, we have
\begin{equation} \label{1.1.3}
h(x, s) \leq |\lambda|\,\|\beta\|_{L^{\infty}(\Omega)} s^q + \lambda_0 s^{p-1} + f (s)\,.
\end{equation}
As a consequence of this and of \eqref{f2origin} we get
\begin{equation}\label{liminfAAA}
\liminf_{s\to 0^+} \frac{h(x,s)}{s^{p-1}}
\leq \liminf_{s\to 0^+}\Big(\lambda|\,\|\beta\|_{L^{\infty}(\Omega)} s^{q-p+1} + \tilde\lambda_0 + \frac{f (s)}{s^{p-1}}\Big)= \tilde\lambda_0 -\ell_0 < 0
\end{equation}
uniformly a.e. $x\in \Omega$, thanks to the choice of $q$. Thus, there exists a sequence $\{s_j\}_j$ converging to 0 as $j\to +\infty$ such that $h(x,s_j) < 0$ for $j\in \NN$ large enough and uniformly a.e. $x\in \Omega$.  Thus, by using the continuity of $s\mapsto h(\cdot, s)$, there exist two sequences $\{\delta_j\}_j , \{\eta_j\}_j$ such that $0 < \eta_{j+1} < \delta_j < s_j < \eta_j, \displaystyle\lim_{j\to +\infty} \eta_j = 0$, and $$h(x,s) \leq 0,$$ for a.e. $x\in \Omega$ and all $s\in [\delta_j,\eta_j]$ and $j$ large enough.
Therefore, hypothesis \eqref{h3origin} holds. Arguing as in the proof of assertion~$a)$ and applying Theorem~\ref{thauxorigin} we get $c)$.

\bigskip

Finally, let us consider the case when $0<q<p-1$. In this setting the strategy will consist in applying Theorem~\ref{thauxiliary} to problem~\eqref{auxiliary} with a suitable choices of $K$ and $h$. At this purpose, let $\tilde{\lambda}_0 \in (0, \ell_0)$, where $\ell_0>0$ is given in assumption~\eqref{f2origin}, and let
\begin{equation} \label{1.1.1BIS}
K(x) := \tilde{\lambda}_0 \,\, \,\,\,{\rm and} \,\,\,\,\, h(x, s, \lambda) := \lambda \beta(x)s^q+\tilde{\lambda}_0 s^{p-1} + f (s),
\end{equation}
a.e. $x\in \Omega$, $s\geq 0$ and $\lambda \in \RR$.

Using the fact that
$$h(x,s,0)=\tilde{\lambda}_0s^{p-1}+f(s)\,,$$
and arguing as in \eqref{1.1.3}--\eqref{liminfAAA},
we get that there exist the sequences $\{\delta_j\}_j , \{\eta_j\}_j, \{s_j\}_j$ and $\{\lambda_j\}_j$ such that $\lambda_j>0$,
\begin{equation}\label{staraggiunta}
0 < \eta_{j+1} < \delta_j < s_j < \eta_j<1, \displaystyle\lim_{j\to +\infty} \eta_j = 0\,,
\end{equation}
and
\begin{equation}\label{hjnegativo}
h(x,s, \lambda) \leq 0
\end{equation}
a.e. $x\in \Omega$, for all $s\in [\delta_j,\eta_j]$, $\lambda\in [-\lambda_j, \lambda_j]$ and $j\in \NN$ large enough.

For any $j\in \NN$, let $h_j:\Omega \times [0,+\infty) \times [-\lambda_j, \lambda_j]\to \RR$ be the function defined by
\begin{equation} \label{3.1.1TER}
h_j(x,s, \lambda)=h(x,\tau_{\eta_j}(s), \lambda),
\end{equation}
and
$$H_j(x, s, \lambda) :=\displaystyle \int^s_0 h_j(x, t, \lambda) dt$$
for a.e. $x\in \Omega$, $s\geq 0$ and $\lambda \in [-\lambda_j, \lambda_j]$.

Let us prove that $K$ given in \eqref{1.1.1BIS}  and $h_j$ satisfy all the assumptions of Theorem~\ref{thauxiliary}. Of course, \eqref{K} and \eqref{condh} are trivially verified, also thanks to \eqref{f0}. Moreover, the regularity of $h$ and the continuity of $\tau_\eta$ show that $h_j$ is a Carath\'eodory function. Also, thanks to \eqref{3.1.1TER}, \eqref{tau}, the continuity of $s\mapsto h(\cdot, s, \cdot)$ and the Weierstrass Theorem give that $h_j$ satisfies \eqref{condhbis}. Finally, \eqref{hjnegativo} and \eqref{3.1.1TER} yield \eqref{hnegativo} for $j$ large enough.
Hence, $h_j$ satisfies all the assumptions of Theorem~\ref{thauxiliary} for $j$ large.

Now, for any $j\in \NN$ let $\mathcal E_{j, \lambda}$ be the energy functional
\begin{equation}\label{Ejlambda}
\mathcal E_{j, \lambda}:=\mathcal E_{K, h_j(\cdot, \cdot,\lambda)}\,,
\end{equation}
where $\mathcal E_{K,\,h_j(\cdot, \cdot,\lambda)}$ is the functional given in \eqref{EKh}, with $h=h_j(\cdot, \cdot,\lambda)$.
By Theorem~\ref{thauxiliary}
we get that, for $j$ sufficiently large and provided $|\lambda|\leq \lambda_j$, there exists $u_{j, \lambda} \in W^{\eta_j}$ such that
\begin{equation} \label{3.1.2ter}
\min_{u\in W^{\eta_j}}\mathcal E_{j, \lambda}(u)=\mathcal E_{j, \lambda}(u_{j, \lambda})
\end{equation}
\begin{equation} \label{3.1.3ter}
u_{j, \lambda} (x)\in [0,\delta_j] {\rm \,\,\,\,\, for \,\, a.e.} \,\, x\in \Omega,
\end{equation}
and
\begin{equation} \label{3.1.4ter}
u_{j, \lambda}\,\,\, \mbox{is a non-negative weak solution of} \,\, (P^K_{h_j(\cdot, \cdot, \lambda)})\,.
\end{equation}
Since for $j$ sufficiently large
\begin{equation}\label{ADDujlambda}
0\leq u_{j, \lambda}(x)\leq \delta_j<\eta_j
\end{equation}
a.e. $x\in \Omega$ by \eqref{staraggiunta} and \eqref{3.1.3ter}, we get
$$h_j(x, u_{j,\lambda}(x), \lambda)=h(x, u_{j,\lambda}(x), \lambda)\,,$$
so that by \eqref{1.1.1BIS} it is easily seen that $u_{j, \lambda}$ is a non-negative weak solution of problem~\eqref{problem}, provided $j$ is large and $|\lambda|\leq\lambda_j$.

It remains to prove that for any $k\in \NN$ problem~\eqref{problem} admits at least $k$ distinct solutions, for suitable values of $\lambda$.
At this purpose, first of all note that, thanks to the choices of $K$ and $h_j$ and \eqref{ADDujlambda}, the functional~$\mathcal E_{j,\lambda}$ is given by
\begin{equation}\label{VV}
\begin{aligned}
\mathcal{E}_{j,\lambda}(u) & = \int_\Omega a(x,\nabla u(x))\,dx-\frac{\lambda}{q+1}\int_\Omega \beta(x)|u(x)|^{q+1}\,dx - \int_\Omega F(u(x))\,dx\\
& =\mathcal{E}_{j,0}(u) -\frac{\lambda}{q+1}\int_\Omega \beta(x)|u(x)|^{q+1}\,dx
\end{aligned}
\end{equation}
for any $u\in W^{1,p}_0(\Omega)$.

We claim that there exists an increasing sequence $\{\theta_j\}_j$ such that $\theta_j< 0$, ${\displaystyle \lim_{j\to +\infty}\theta_j=0}$ and
\begin{equation}\label{Eitetai}
\theta_{j-1}<\mathcal E_{j,0}(u_{j,0})<\theta_j
\end{equation}
for $j\geq j^*$, with $j^*\in \NN$.

First, note that the function
$$(x, s)\mapsto h(x,s,0)=\tilde \lambda_0s^{p-1}+f(s)$$
verifies all the assumptions of Theorem~\ref{thauxorigin}, in particular thanks to \eqref{f1origin}. Hence, arguing as in the proof of Theorem~\ref{thauxorigin} (see, in particular, formulas~\eqref{3.1.7}--\eqref{BB}) we get that there exist $\ell > 0$ and $\zeta\in (0,\eta_1)$ such that
\begin{equation} \label{3.1.7bis}
F(s) \geq -\, \ell s^p\,\,\, \mbox{for all} \,\,\, s\in (0,\zeta)
\end{equation}
and that
there is a sequence $\{\tilde s_j\}_j$ such that $0<\tilde s_j \to 0$ as $j\to +\infty$  such that
for any $L>0$
\begin{equation}\label{3.1.9bis}
F(s_j)> L s_j^p
\end{equation}
for $j\in \NN$ large enough.
Also, since $\delta_j\searrow 0$ as $j\to +\infty$, we can choose a subsequence of $\{\delta_j\}_j$, still denoted by $\{\delta_j\}_j$, such that
\begin{equation}\label{BBbis}
\tilde s_j\leq \delta_j
\end{equation}
for all $j\in \NN$.

Now, let us fix $j\in \NN$ sufficiently large and let
$$\tilde z_j:=z_{\tilde s_j}\in W_0^{1,p}(\Omega)$$
be the function defined as in \eqref{2.1.1} with $s=\tilde s_j$.
Arguing as in \eqref{Ejzj}, by \eqref{3.1.7bis} and \eqref{3.1.9bis} we get that for $j$ large enough
\begin{equation}\label{cj}
\mathcal E_{j,0}(u_{j,0})\leq \mathcal E_{j,0}(\tilde z_j)<-C_1\tilde s_j^p=:c_j<0\,,
\end{equation}
for a suitable positive constant $C_1$.
Also, as in \eqref{claim2proof} and taking into account the continuity of $f$ and \eqref{ADDujlambda}, we have that, for $j$ sufficiently large
\begin{equation}\label{dj}
\begin{aligned}
\mathcal E_{j,0}(u_{j,0}) & \geq -\int_\Omega F(u_{j,0}(x))\,dx\\
& \geq -\int_\Omega \int_0^{u_{j,0}(x)}|f(s)|\,ds\,dx\\
& \geq -\int_\Omega \int_0^{\delta_j}|f(s)|\,ds\,dx\\
& \geq -C_2\delta_j=:d_j<0\,,
\end{aligned}
\end{equation}
where $C_2$ is a positive constant.

Note that $\{c_j\}_j$ and $\{d_j\}_j$ are such that $d_j<c_j<0$ for any $j\in \NN$ and ${\displaystyle \lim_{j\to +\infty}c_j=\lim_{j\to +\infty}d_j=0}$, since $\{\delta_j\}_j$ does, thanks to \eqref{staraggiunta}. Thus, we can extract two subsequences, still denoted by $\{c_j\}_j$ and $\{d_j\}_j$, such that the above properties hold true and the sequences $\{c_j\}_j$ and $\{d_j\}_j$ are non-decreasing. Now, we define
$$\theta_j:=\left\{\begin{array}{ll}
c_j & \mbox{if $j\in \NN$ is even}\\
d_j & \mbox{if $j\in \NN$ is odd}\,.
\end{array}
\right.$$
As a consequence of this definition and of \eqref{cj} and \eqref{dj} we have that for $i$ large enough
$$\theta_{2i-1}= d_{2i-1}\leq d_{2i}<\mathcal E_{2i,0}(u_{2i,0})<c_{2i}=\theta_{2i}\,,$$
which, putting $2i=j$ gives \eqref{Eitetai}. Hence, the claim is proved.
Note that $\theta_j$ is independent of $\lambda$, since $c_j$ and $d_j$ do.

Now, for any $j\geq j^*$ let
\begin{equation}\label{}
\lambda'_j:=\frac{(q+1)(\mathcal E_{j,0}(u_{j,0})-\theta_{j-1})}{(\|\beta\|_{L^\infty(\Omega)}+1)\mathcal L(\Omega)}
\qquad \lambda''_j:=\frac{(q+1)(\theta_j-\mathcal E_{j,0}(u_{j,0}))}{\|\beta\|_{L^1(\Omega)}+1}\,.
\end{equation}
Note that $\lambda'_j$ and $\lambda''_j$ are strictly positive , thanks to \eqref{Eitetai}, and they are independent of $\lambda$.

Now, for any fixed $k\in \NN$, let
$$\Lambda_k:=\min\Big\{\lambda_{j^*+1}, \dots, \lambda_{j^*+k}, \lambda'_{j^*+1}, \dots, \lambda'_{j^*+k}, \lambda''_{j^*+1}, \dots, \lambda''_{j^*+k}\Big\}\,.$$
Of course, $\Lambda_k>0$ is independent of $\lambda$\,.
Also, if $|\lambda|\leq\Lambda_k$, then $|\lambda|\leq\lambda_j$ for any $j=j^*+1, \dots, j^*+k$\,.
As a consequence of this, for any $\lambda\in \RR$ with $|\lambda|\leq\Lambda_k$
$$u_{j,\lambda}\,\,\, \mbox{is a non-negative weak solution of problem}~\eqref{problem}$$
for any $j=j^*+1, \dots, j^*+k$\,.

Let us show that these solutions are distinct. At this purpose, note that $u_{j,\lambda}\in W^{\eta_j}$ by \eqref{ADDujlambda} and so
\begin{equation}\label{VVV}
\mathcal E_{j, 0}(u_{j, 0})=\min_{u\in W^{\eta_j}}\mathcal E_{j, 0}(u)\leq \mathcal E_{j, 0}(u_{j, \lambda})\,.
\end{equation}
By \eqref{VV} and \eqref{VVV}, for any $\lambda$ such that $|\lambda|\leq\Lambda_k$ we get
\begin{equation}\label{Ej1}
\begin{aligned}
\mathcal E_{j, \lambda}(u_{j, \lambda}) & = \mathcal E_{j, 0}(u_{j, \lambda})-\frac{\lambda}{q+1}\int_\Omega \beta(x)|u_{j, \lambda}(x)|^{q+1}\,dx\\
& \geq \mathcal E_{j, 0}(u_{j, 0})-\frac{|\lambda|}{q+1}\|\beta\|_{L^\infty(\Omega)}\eta_j^{q+1} \mathcal L(\Omega)\\
& \geq \mathcal E_{j, 0}(u_{j, 0})-\frac{\Lambda_k}{q+1}\|\beta\|_{L^\infty(\Omega)}\mathcal L(\Omega)\\
& \geq \mathcal E_{j, 0}(u_{j, 0})-\frac{\lambda'_j}{q+1}\|\beta\|_{L^\infty(\Omega)}\mathcal L(\Omega)\\
& =\theta_{j-1}\,,
\end{aligned}
\end{equation}
for any $j=j^*+1, \dots, j^*+k$\,, thanks to \eqref{staraggiunta}, \eqref{ADDujlambda}, the choice of $\Lambda_k$ and the definition of $\lambda'_j$.

On the other hand, by \eqref{cj}, \eqref{VV} and using the fact that $\|\tilde z_j\|_{L^\infty(\Omega)}=\tilde s_j\leq\delta_j< \eta_j<1$ (see \eqref{staraggiunta} and \eqref{BBbis}), for any $\lambda$ with $|\lambda|\leq\Lambda_k$ we deduce that
\begin{equation}\label{Ej2}
\begin{aligned}
\mathcal E_{j, \lambda}(u_{j, \lambda}) & =\min_{u\in W^{\eta_j}}\mathcal E_{j, \lambda}(u)\\
& \leq \mathcal E_{j, \lambda}(\tilde z_j)\\
& = \mathcal E_{j, 0}(\tilde z_j)-\frac{\lambda}{q+1}\int_\Omega \beta(x)|\tilde z_j(x)|^{q+1}\,dx\\
& \leq \mathcal E_{j, 0}(\tilde z_j)-\frac{\lambda}{q+1}\int_{\{x\in \Omega : \lambda \beta(x)<0\}} \beta(x)|\tilde z_j(x)|^{q+1}\,dx\\
& \leq \mathcal E_{j, 0}(\tilde z_j)-\frac{\lambda}{q+1}\int_{\{x\in \Omega : \lambda \beta(x)<0\}} \beta(x)\,dx\\
& \leq \mathcal E_{j, 0}(\tilde z_j)+\frac{|\lambda|}{q+1}\int_{\{x\in \Omega : \lambda \beta(x)<0\}} |\beta(x)|\,dx\\
& \leq \mathcal E_{j, 0}(\tilde z_j)+\frac{\lambda}{q+1}\int_\Omega |\beta(x)|\,dx\\
& \leq \mathcal E_{j, 0}(\tilde z_j)+\frac{\Lambda_k}{q+1}\|\beta\|_{L^1(\Omega)}\\
& \leq \mathcal E_{j, 0}(\tilde z_j)+\frac{\lambda''_j}{q+1}\|\beta\|_{L^1(\Omega)}\\
& =\theta_j\,,
\end{aligned}
\end{equation}
for any $j=j^*+1, \dots, j^*+k$\,, again thanks to the choice of $\Lambda_k$ and the definition of $\lambda''_j$.

Hence, by \eqref{Ej1}, \eqref{Ej2} and the properties of $\{\theta_j\}_j$ we deduce that for any $j=j^*+1, \dots, j^*+k$
\begin{equation}\label{Eifinale}
\theta_{j-1}<\mathcal E_{j, \lambda}(u_{j, \lambda})<\theta_j<0\,,
\end{equation}
which yields that
$$\mathcal E_{1, \lambda}(u_{1, \lambda})< \dots < \mathcal E_{k, \lambda}(u_{k, \lambda})<0\,,$$
that is the solutions $\Big\{u_{1, \lambda}, \dots, u_{k, \lambda}\Big\}$ are all distinct and non-trivial, provided $|\lambda|\leq\Lambda_k$.

Finally, let us estimate the $W^{1,p}_0$--norm of $u_{j,\lambda}$. For this, by \eqref{anorma}, \eqref{staraggiunta}, \eqref{ADDujlambda}, \eqref{VV} and \eqref{Eifinale} we have that for any $j=j^*+1, \dots, j^*+k$ and $|\lambda|\leq\Lambda_k$
$$\begin{aligned}
\frac{\Gamma_1}{p}\|u_{j,\lambda}\|_{W^{1,p}_0(\Omega)}^p & \leq \mathcal E_{j,\lambda}(u_{j,\lambda})+\frac{\lambda}{q+1}\int_\Omega \beta(x)|u_{j,\lambda}(x)|^{q+1}\,dx + \int_\Omega F(u_{j,\lambda}(x))\,dx\\
& < \theta_j+\frac{|\lambda|}{q+1}\|\beta\|_{L^\infty(\Omega)}\delta_j^{q+1}+ \int_\Omega \int_0^{\delta_j}|f(s)|\,ds\,dx\\
& < \frac{\Lambda_k}{q+1}\|\beta\|_{L^\infty(\Omega)}\delta_j + \bar C\delta_j\,,
\end{aligned}$$
for a suitable positive constant $\bar C$, that is
$$\|u_{j,\lambda}\|_{W^{1,p}_0(\Omega)}\leq \tilde C\delta_j^{1/p}\,,$$
where $\tilde C>0$\,.
Since $\delta_j\to 0$ as $j \to +\infty$, without loss of generality, we may assume that
\begin{equation}\label{sceltadj}
\delta_j\leq \min\{\tilde C^{-p}, 1\} 1/j^p
\end{equation}
and this gives
$$\|u_{j,\lambda}\|_{W^{1,p}_0(\Omega)}\leq 1/j$$
for any $j=j^*+1, \dots, j^*+k$, provided $|\lambda|\leq\Lambda_k$,
which is the desired assertion. Finally, by \eqref{ADDujlambda} and \eqref{sceltadj} it is easilty seen that
$$\|u_{j,\lambda}\|_{L^\infty(\Omega)}\leq 1/j^p<1/j$$
for any $j=j^*+1, \dots, j^*+k$, provided $|\lambda|\leq\Lambda_k$
This completes the proof of Theorem~\ref{thorigin}.


\section{Oscillation at infinity}\label{sec:infty}
This section is devoted to the study of problem~\eqref{problem} in the case when $f$ oscillates at infinity.

In order to prove Theorem~\ref{thinfty} we apply some techniques used in the previous section. However, for completeness, we give all the details.

Here, we consider again problem~\eqref{auxiliary}, under the following assumptions on the function~$h$:
\begin{equation}\label{h1infty}
\mbox{for any}\,\,\, s\geq 0\,\,\, \sup_{t\in [0,s]} |h(\cdot, t)|\in L^{\infty}(\Omega)\,;
\end{equation}
\begin{equation}\label{h3infty}
\begin{aligned}
& \qquad \mbox{there exist two sequences}\,\,\, \{\delta_j\}_j\,\,\mbox{and}\,\, \{\eta_j\}_j\,\,\, \mbox{with}\,\,\, 0 < \delta_j < \eta_j < \delta_{j+1}\,\,\,  \mbox{and}\\
& \displaystyle\lim_{j\to +\infty} \delta_j = +\infty\,\,\, \mbox{such that}\,\,\, h(x, s) \leq 0\,\,\, \mbox{for a.e.}\,\,\, x\in \Omega\,\,\, \mbox{and for all}\,\,\, s\in [\delta_j,\eta_j], j\in \NN\,;
\end{aligned}
\end{equation}
\begin{equation}\label{h2infty}
\displaystyle -\infty< \liminf_{s\to +\infty}\frac{H(x,s)}{s^{p}}\leq\displaystyle\limsup_{s\to +\infty} \displaystyle\frac{H(x,s)}{s^p}=+\infty\,\,\,\mbox{uniformly for a.e.}\,\,\, x\in \Omega\,,
\end{equation}
where $H$ is the function given in \eqref{H}.

In this context our existence result for problem~\eqref{auxiliary} is given by the following theorem:
\begin{theorem}\label{thauxinfty}
Let $\Omega\subset \RR^N$, $N \geq 3$, be a bounded domain with smooth boundary, $\lambda\in \RR$ and let $A:\overline \Omega \times \RR^N\to \RR^N$ and $a:\overline\Omega\times \RR^N\to \RR$ be two functions satisfying \eqref{Acontinua}--\eqref{potenziale} and \eqref{acontinua}--\eqref{Aconvex}, respectively. Moreover, assume that $K:\Omega \to \RR$ satisfies \eqref{K} and $h:\Omega \times [0,+\infty)\to \RR$ is a Carath\'eodory function verifying \eqref{condh} and \eqref{h1infty}--\eqref{h2infty}.

Then, there exists a sequence $\{u_j\}_j \subset W_0^{1,p}(\Omega)$ of distinct non-negative weak solutions of problem~\eqref{auxiliary} such that
\begin{equation} \label{4.1.0}
\lim_{j\to +\infty} \|u_j\|_{L^{\infty}(\Omega)} = +\infty.
\end{equation}
\end{theorem}
\begin{proof}
Here we consider again the function~$h_j$ and the functional~$\mathcal E_j$ defined in \eqref{3.1.1} and \eqref{funzionaleE}, respectively.
Taking into account hypotheses \eqref{h1infty} and \eqref{h3infty}, it is easily seen that $h_j$ fulfills the assumptions of Theorem~\ref{thauxiliary} for any $j\in \NN$. Thus, for every $j\in \NN$, there is an element $u_j \in W^{\eta_j}$ such that
\begin{equation} \label{4.1.2}
u_j\,\,\, \mbox{is the minimum point of the functional}\,\,\, \mathcal{E}_j\,\, \mbox{on}\,\, W^{\eta_j},
\end{equation}
\begin{equation} \label{4.1.3}
u_j (x)\in [0,\delta_j] {\rm \,\,\,\,\, for \,\, a.e.} \,\, x\in \Omega,
\end{equation}
and
\begin{equation} \label{4.1.4}
u_j \,\,\, \mbox{is a non-negative weak solution of} \,\, (P^K_{h_j}).
\end{equation}

Arguing as in the proof of Theorem~\ref{thauxorigin} and taking into account the definition of $h_j$, \eqref{h3infty} and \eqref{4.1.3}, it is easily seen that
$$h_j(x,u_j(x))=h(x,\tau_{\eta_j}(u_j(x)))=h(x,u_j(x))\,,$$
so that, by \eqref{4.1.4}, $u_j$ is also a non-negative weak solution of problem~\eqref{auxiliary}.

In order to get the assertion of Theorem~\ref{thauxinfty}, we need to show that there are infinitely many distinct elements in the sequence $\{u_j\}_j$. To this end, first of all we claim that, up to a subsequence,
\begin{equation} \label{4.1.5}
\displaystyle \lim_{j\to +\infty} \mathcal{E}_j(u_j)=-\infty.
\end{equation}

For this, note that, by \eqref{h2infty}, there exist $\ell > 0$ and $\zeta > 0$ such that
\begin{equation} \label{4.1.6}
\mathop {\rm ess\,inf}_{x\in \Omega} H(x, s) \geq -\, \ell s^p\,\,\, \mbox{for all} \,\,\, s>\zeta
\end{equation}
and there exists a sequence $\{s_j\}_j$ such that $\displaystyle\lim_{j\to +\infty} s_j=+\infty$ and
$$\displaystyle\limsup_{j\to +\infty} \displaystyle\frac{H(x,s_j)}{s_j^p}=+\infty\,,$$
that is, for any $L>0$
\begin{equation} \label{4.1.8}
\mathop {\rm ess\,inf}_{x\in \Omega} H(x, s_j) > L s_j^p
\end{equation}
for $j\in \NN$ sufficiently large.

Since $\delta_j \nearrow +\infty$ by \eqref{h3infty}, we can choose a subsequence of $\{\delta_j\}_j$, still denoted by $\{\delta_j\}_j$, such that
\begin{equation}\label{BBB}
s_j \leq \delta_j
\end{equation}
for all $j\in \NN$. Let us fix $j\in \NN$ and let
$$z_j:=z_{s_j}\in W_0^{1,p}(\Omega)$$
be the function from \eqref{2.1.1} with $s=s_j$. Then, $z_j\in W^{1,p}_0(\Omega)$ and $\|z_j\|_{L^\infty(\Omega)}=s_j$, so that, by \eqref{h3infty} and \eqref{BBB}, $0\leq z_j(x)\leq \delta_j<\eta_j$ a.e. $x\in \Omega$. Taking into account \eqref{box2}, \eqref{2.1.2}, \eqref{4.1.6} and \eqref{4.1.8}, we have
\begin{equation}\label{Evv}
\begin{aligned}
\mathcal{E}_j(z_j) & = \displaystyle \int_\Omega a(x,\nabla z_j(x))dx+\frac{1}{p}\int_\Omega K(x)|z_j(x)|^pdx - \int_\Omega H_j(x,z_j(x)) dx\\
& \leq \displaystyle  C(r,p,N)\frac{\Gamma_2}{p} s_j^p + \frac{1}{p}\int_\Omega K(x)|z_j(x)|^pdx - \displaystyle \int_{B(x_0,r/2)} H(x, s_j)dx\\
& \qquad \displaystyle - \int_{(B(x_0,r) \setminus B(x_0,r/2))\cap\{z_j>\,\zeta\}} H\big(x, z_j(x)\big)dx\\
& \qquad \displaystyle - \int_{(B(x_0,r) \setminus B(x_0,r/2))\cap\{z_j\leq \,\zeta\}} H\big(x, z_j(x)\big)dx\\
&\displaystyle \leq \Big(C(r,p,N)\frac{\Gamma_2}{p} +  \frac{\|K\|_{L^{\infty}(\Omega)} \mathcal{L}(\Omega)}{p} - L(r/2)^N \omega_N + \ell \mathcal{L}(\Omega) \Big) s_j^p\\
& \qquad \displaystyle + \Big\| \sup_{s\in [0,\,\zeta]} |h(\cdot, s)| \Big\|_{L^{\infty}(\Omega)} \mathcal{L}(\Omega)\zeta.
\end{aligned}
\end{equation}
Choosing $L>0$ sufficiently large, so that
$$L(r/2)^N \omega_N > C(r,p,N)\frac{\Gamma_2}{p} +  \frac{\|K\|_{L^{\infty}(\Omega)} \mathcal{L}(\Omega)}{p} + \ell \mathcal{L}(\Omega)\,,$$
and taking into account that $\displaystyle\lim_{j\to +\infty} s_j=+\infty$, by \eqref{Evv} we get
\begin{equation} \label{4.1.9}
\displaystyle \lim_{j\to +\infty}\mathcal{E}_j(z_j)= -\infty\,.
\end{equation}

On the other hand, using \eqref{4.1.2}, we have
$$\mathcal{E}_j(u_j) = \displaystyle \min_{u\in W^{\eta_j}}\mathcal{E}_j(u)\\
\leq \mathcal{E}_j (z_j)\,,$$
so that, by this and \eqref{4.1.9} it easily follows claim~\eqref{4.1.5}.

As a consequence of \eqref{4.1.5} we get that the sequence $\{u_j\}_j$ has infinitely many distinct elements (and, in particular, $u_j\not \equiv 0$ in $\Omega$, being $\mathcal E_j(0)=0$).
Indeed, let us assume that in the sequence $\{u_j\}_j$ there is only a finite number of elements, say $\{u_1,\dots, u_k\}$ for some $k \in \NN$. Consequently, the sequence $\{\mathcal{E}_j(u_j)\}_j$ reduces to at most the finite set $\{\mathcal{E}_1(u_1),\dots, \mathcal{E}_k(u_k)\}$ and this fact contradicts relation \eqref{4.1.5}.
Hence, problem~\eqref{auxiliary} admits infinitely many distinct weak solutions.

Now, we prove \eqref{4.1.0}, arguing by contradiction. We assume that, up to a subsequence, still denoted by $\{u_j\}_j$ , there exists $L>0$ such that
\begin{equation}\label{ujbounded}
\|u_j\|_{L^{\infty}(\Omega} \leq L
\end{equation}
for all $j\in \NN$\,.

Since $\eta_j\to +\infty$ as $j\to +\infty$, for $j$ large enough, say $j\geq j^*$, with $j^*\in \NN$, we have that $\eta_j\geq L$. As a consequence of this and \eqref{ujbounded}, we deduce that
\begin{equation}\label{NNN}
u_j\in W^{\eta_{{j^*}}}\,\,\,\,  \mbox{for any}\,\,\,\, j\geq j^*\,.
 \end{equation}
Here we used also the fact that the sequence $\{\eta_j\}_j$ is increasing by \eqref{h3infty}.

Also, note that, as a consequence of the monotonicity of $\{\eta_j\}_j$, it is easy to see that when $u\in W^{\eta_j}$ and $j<k$, then $u\in W^{\eta_k}$, that is
\begin{equation}\label{Winclusione}
W^{\eta_j}\subseteq W^{\eta_k}
\end{equation}
and this implies that a.e. $x\in \Omega$
\begin{equation}\label{Hjk}
\begin{aligned}
H_j(x, u(x))& =\int_0^{u(x)} h(x, \tau_{\eta_j}(t))dt\\
& =\int_0^{u(x)} h(x, t)dt\\
& =\int_0^{u(x)} h(x, \tau_{\eta_k}(t))dt\\
& =H_k(x, u(x))\,,
\end{aligned}
\end{equation}
provided $u\in W^{\eta_j}$\,.

Furthermore, the sequence $\{\mathcal{E}_j(u_j)\}_j$ is non-increasing. Indeed, for $j<k$, by \eqref{Winclusione} and \eqref{Hjk} we have
\begin{equation}\label{Enonincreasing}
\begin{aligned}
\mathcal{E}_j(u_j) & = \displaystyle \min_{u\in W^{\eta_{j}}}\mathcal{E}_j(u)\\
& = \displaystyle \min_{u\in W^{\eta_{j}}}\mathcal{E}_k(u)\\
& \geq \displaystyle \min_{u\in W^{\eta_{k}}}\mathcal{E}_k(u)\\
& =\mathcal{E}_k (u_k).
\end{aligned}
\end{equation}

Then, by \eqref{NNN}--\eqref{Enonincreasing}, for any $j\geq j^*$ we get
$$\begin{aligned}
\mathcal{E}_{j^*}(u_{j^*}) & \geq \mathcal E_j(u_j)\\
&\geq \displaystyle \min_{u\in W^{\eta_{j^*}}}\mathcal{E}_j(u)\\
& = \displaystyle \min_{u\in W^{\eta_{j^*}}}\mathcal{E}_{j^*}(u)\\
& =\mathcal{E}_{j^*}(u_{j^*})\,.
\end{aligned}$$
 As a consequence,
$$\mathcal{E}_j(u_j) =\mathcal{E}_{j^*}(u_{j^*})\,\,\, \mbox{for all}\,\, j\geq j^*\,.$$
This fact contradicts \eqref{4.1.5}. Hence, $\|u_j\|_{L^\infty(\Omega)}\to +\infty$ as $j\to +\infty$ and this concludes the proof of Theorem~\ref{thauxinfty}\,.
\end{proof}

Requiring the following extra condition on the function $h$
\begin{equation} \label{1.3.a}
\sup_{s\in [0,+\infty)} \frac{|h (x,s)|}{1+ s^{p^*-1}} <+\infty\,\,\, \mbox{uniformly a.e.}\,\, x\in \Omega,
\end{equation}
where $p^*$ is the critical Sobolev exponent given in \eqref{pstar},
we have the next result:
\begin{corollary}\label{corinfty}
Let all the assumptions of Theorem~\ref{thauxinfty} be satisfied. In addition assume that \eqref{1.3.a} holds true.

Then, $$\lim_{j\to +\infty} \|u_j\|_{W^{1,p}_0(\Omega)}=+\infty\,,$$
where $\{u_j\}_j$ is the sequence of distinct weak solutions of problem~\eqref{auxiliary}, given by Theorem~\ref{thauxinfty}.
\end{corollary}
\begin{proof}
We argue by contradiction and we assume that, up to a subsequence, still denoted by $\{u_j\}_j$, there exists $L>0$ such that for any $j\in \NN$
\begin{equation}\label{ujL}
\|u_j\|_{W^{1,p}_0(\Omega)}\leq L\,.
\end{equation}

Hence, for any $j\in \NN$ by \eqref{anorma} and \eqref{1.3.a} we have
\begin{equation}\label{Elimitato}
\begin{aligned}
|\mathcal E_j(u_j)|& \leq \frac{\Gamma_2}{p}\|u_j\|_{W^{1,p}_0(\Omega)}^p+\|K\|_{L^\infty(\Omega)}\|u_j\|_{L^p(\Omega)}^p+
C_1\int_\Omega \int_0^{u_j(x)}(1+|s|^{p^*-1})\,ds\,dx\\
& \leq \frac{\Gamma_2}{p}\|u_j\|_{W^{1,p}_0(\Omega)}^p+\|K\|_{L^\infty(\Omega)}\|u_j\|_{L^p(\Omega)}^p+
C_2\|u_j\|_{L^1(\Omega)}+C_3\|u_j\|_{L^{p^*}(\Omega)}^{p*}\\
\end{aligned}
\end{equation}
for suitable positive constants $C_1$, $C_2$ and $C_3$.
Since the Sobolev space $W_0^{1,p}(\Omega)$ is continuously embedded into $L^\gamma(\Omega)$ for any $\gamma\in [1, p^*]$, by \eqref{ujL} and \eqref{Elimitato} we easily obtain that the sequence $\{\mathcal E_j(u_j)\}_j$ is bounded in $\RR$. This contradicts \eqref{4.1.5} and this ends the proof of Corollary~\ref{corinfty}.
\end{proof}

\subsection{Proof of Theorem~\ref{thinfty}}\label{subsec:proofinfty}
We use some techniques developed in the proof of Theorem~\ref{thorigin}.  The main idea consists in applying Theorem~\ref{thauxiliary} and Theorem~\ref{thauxinfty} to problem~\eqref{auxiliary} with a suitable choice for the functions~$K$ and $h$ appearing in the equation.

In the case when $q=p-1$ and $\ell_\infty\in (0, +\infty)$ we fix $\lambda \in \RR$ such that $\lambda \beta(x) < \lambda_{\infty}$ a.e. $x\in \Omega$ for some $\lambda_{\infty}\in (0,\ell_\infty)$. In this setting we take $\tilde \lambda_\infty\in (\lambda_\infty, \ell_\infty)$ and
\begin{equation}\label{chooseq=p-1}
K(x) := \tilde{\lambda}_{\infty} - \lambda \beta(x)\,\,\, \,\,\,\mbox{and}\,\,\, \,\,\, h(x, s) := \tilde{\lambda}_{\infty} s^{p-1} + f (s)\,,
\end{equation}
for a.e. $x\in \Omega$ and  $s\geq 0$\,, in Theorem~\ref{thauxinfty}.
With the same arguments used in the proof of Theorem~\ref{thorigin}, it is easily seen that $K$ and $h$ satisfy the assumptions of Theorem~\ref{thauxinfty} (here we use also the fact that $f(0)=0$ by assumption) and this gives the assertion of Theorem~\ref{thinfty}.

\medskip

In the case when $q=p-1$ and $\ell_\infty=+\infty$ we take $\lambda \in \RR$ and we use Theorem~\ref{thauxinfty} with
\begin{equation}\label{chooseq=p-1infty}
K(x) := \tilde{\lambda}_{\infty}\,\,\, \,\,\,\mbox{and}\,\,\, \,\,\, h(x, s) := \Big(\lambda \beta(x)+\tilde{\lambda}_{\infty}\Big) s^{p-1} + f (s)\,,
\end{equation}
for a.e. $x\in \Omega$ and  $s\geq 0$\,. The arguments are the same of the ones used in the previous case.

\medskip

In the case when $0<q<p-1$ we choose $K$ and $h$ in Theorem~\ref{thauxinfty} as follows:
\begin{equation}\label{chooseq<p-1}
K(x) := \tilde\lambda_{\infty} \,\,\,\,\,{\rm and} \,\,\,\,\, h(x, s) := \lambda \beta(x) s^{q} + \tilde\lambda_{\infty} s^{p-1} + f (s)\,,
\end{equation}
for a.e. $x\in \Omega$ and $s\geq 0$\,, where $\tilde \lambda_\infty\in (0, \ell_\infty)$, and we argue as above.

In the case when $q>p-1$ the strategy will consist in applying Theorem~\ref{thauxiliary} to problem~\eqref{auxiliary} with a suitable choices of $K$ and $h$. At this purpose, let $\tilde{\lambda}_\infty \in (\lambda_\infty, \ell_\infty)$, where $\ell_\infty>0$ is given in assumption~\eqref{f2infty}, and let
$$K(x) := \tilde{\lambda}_\infty \,\, \,\,\,{\rm and} \,\,\,\,\, h(x, s, \lambda) := \lambda \beta(x)s^q+\tilde{\lambda}_\infty s^{p-1} + f (s)$$
a.e. $x\in \Omega$, $s\geq 0$ and $\lambda \in \RR$. Arguing as in the proof of Theorem~\ref{thorigin} we get the assertion.

\medskip

Finally, when $q>p-1$ we can adapt the arguments used in the proof of Theorem~\ref{thorigin}.
This ends the proof of Theorem~\ref{thinfty}.

\bigskip

As a consequence of Corollary~\ref{corinfty}, we have the following result:
\begin{corollary}\label{corinftyf}
Let $q\leq p-1$ and let all the assumptions of Theorem~\ref{thinfty} be satisfied. In addition, assume that
\begin{equation}\label{1.3.af}
\sup_{s\in [0,+\infty)} \frac{|f (s)|}{1+ s^{p^*-1}} <+\infty,
\end{equation}
where $p^*$ is given in \eqref{pstar}.

Then, $$\lim_{j\to +\infty} \|u_j\|_{W^{1,p}_0(\Omega)}=+\infty\,,$$
where $\{u_j\}_j$ is the sequence of distinct weak solutions of problem~\eqref{problem}, given by Theorem~\ref{thinfty}.
\end{corollary}
\begin{proof}
It is enough to apply Corollary~\ref{corinfty} with $K$ and $h$ given in \eqref{chooseq=p-1} when $q=p-1$, and in \eqref{chooseq<p-1} when $0<q<p-1$.

Let us consider the case when $q=p-1$. Note that $q>0$, being $p>1$. Also, by \eqref{1.3.af}, for $s\in [0,1]$
$$\begin{aligned}
\frac{|h(x,s)|}{1+ s^{p^*-1}}& = \frac{\tilde \lambda_\infty s^{p-1}}{1+ s^{p^*-1}}+\frac{|f(s)|}{1+ s^{p^*-1}}\\
& \leq \tilde \lambda_\infty s^{p-1}+\frac{|f(s)|}{1+ s^{p^*-1}}\\
& \leq \tilde \lambda_\infty +\frac{|f(s)|}{1+ s^{p^*-1}}\\
& \leq \tilde \lambda_\infty + \sup_{s\in [0,+\infty)} \frac{|f (s)|}{1+ s^{p^*-1}} <+\infty\,,
\end{aligned}$$
while for $s>1$ we have
$$\begin{aligned}
\frac{|h(x,s)|}{1+ s^{p^*-1}}& = \frac{\tilde \lambda_\infty s^{p-1}}{1+ s^{p^*-1}}+\frac{|f(s)|}{1+ s^{p^*-1}}\\
& \leq \tilde \lambda_\infty s^{p-p^*}+\frac{|f(s)|}{1+ s^{p^*-1}}\\
& \leq \tilde \lambda_\infty s^{p-p^*} + \sup_{s\in [0,+\infty)} \frac{|f (s)|}{1+ s^{p^*-1}}<+\infty\,,
\end{aligned}$$
since $p<p^*$ and $s^{p-p^*}\to 0$ as $s\to +\infty$.
Thus, in any case \eqref{1.3.a} is satisfied.

In the case when $0<q<p-1$ we argue in a similar way, taking into account that $q<p-1<p^*-1$.
This concludes the proof of Corollary~\ref{corinftyf}.
\end{proof}






\end{document} 